

VULNERABILITY, RESILIENCE AND ADAPTATION TO CLIMATE CHANGE IN THE NORTHWEST MOUNTAINOUS REGIONS OF VIETNAM

A Thesis submitted by

Van Thanh Tran

BSc, MSc

For the award of

Doctor of Philosophy

2023

ABSTRACT

Vietnam is one of the most vulnerable countries to climate change in the world. Within the country, the Northwest Mountainous Regions (NMRs) are considered some of the most vulnerable communities, due to topography and socio-economic factors. Therefore, targeted adaptation is increasingly seen as both a necessary and urgent response. However, the extent and dimensions of vulnerability need to be carefully examined in order to develop effective adaptation pathways. This research aimed to explore vulnerability, resilience, adaptation to climate change in the NMRs, particularly among ethnic communities. Specifically, we aimed to (1) identify which communities are more vulnerable to climate change, (2) identify differences in measures of farmers' household resilience among communities, and (3) investigate the major factors contributing to farmers' adaptation choices in relation to climate change. The case studies are located in the Phu Yen district of Son La province in the NMRs. Data were collected in the field from late 2018 to early 2019. Primary data is from interviews and field observations, as well as insights from local decisionmakers, resource managers, scientists. Climate data, which could be compared to perceptions of climate change, was collected from the Phu Yen meteorological station and the Hydro-Meteorological Data Centre of to 2017. The resulting vulnerability Vietnam (HMDC) from 1961 assessment, based on a set of indicators quantified from respondents' selfreporting, indicates that two of the ethnic communities were, on average, more vulnerable, particularly on livelihood strategies, health, water, housing and productive land, and social networks components when compared to the other two ethnic communities. Our study also reveals that indicators of household livelihood resilience differed between ethnic communities and between genders, with ethnicity being relatively more important than gender in determining outcomes. The research identified a number of areas where changes of institutional and socio-economic factors could increase livelihood resilience scores. There is also evidence of potential benefits in targeted resilience programs. Among ethnic groups, women's responses showed, on average, a lower resilience than those of men. The study found that a high percentage of farmers had noticed changes in the frequency of particular climate attributes and climate hazards, somewhat in line with climate data. To cope with these changes, recorded adaptation strategies applied in the study region included crop management and protection (soil and plant), diversifying crop, finding offfarm jobs, and changing crop varieties. Survey respondents also identified obstacles to adaptation measures, such as lack of credit, lack of family labour, insufficient agricultural inputs, limited farm size, and difficulties in assessing updated weather information. Regression results from binary logistic models reveal that age, gender, farm size, irrigation, extension services, credit availability, level of education and updated climate information have a significant influence on farmers' preference for adaptation strategies regarding climate change and climate hazards.

CERTIFICATION OF THESIS

I, Van Thanh Tran declare that the PhD Thesis entitled "Vulnerability, Resilience and Adaptation to Climate Change in the Northwest Mountainous Regions of Vietnam" is not more than 100,000 words in length including quotes and exclusive of tables, figures, appendices, bibliography, references, and footnotes.

This Thesis is the work of Van Thanh Tran except where otherwise acknowledged, with the majority of the contribution to the papers presented as a Thesis by Publication undertaken by the Student. The work is original and has not previously been submitted for any other award, except where acknowledged.

Date: 20th January 2023

Endorsed by:

Prof. Geoff Cockfield Principal Supervisor

Dr. Duc-Anh An-Vo Associate Supervisor

Prof. Shahbaz Mushtaq Associate Supervisor

Student and supervisors' signatures of endorsement are held at the University.

STATEMENT OF CONTRIBUTION

The articles produced from this study had contributions from the student and supervisory team. The details of the scientific contribution of each author in the publications are provided below:

Paper 1:

Van Thanh Tran, Duc-Anh An-Vo, Geoff Cockfield, and Shahbaz Mushtaq (2021). "Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam". *Sustainability, 13(13), 7106.* https://doi.org/10.3390/su13137106.

The overall contribution of Van Thanh Tran was 60% to the concept development, formal analysis, methodology, investigation, data curation, drafting and revising the final submission; Geoff Cockfield and Duc-Anh An-Vo contributed 20% and 15% respectively, for concept development, formal analysis, editing and provided the comments on the manuscript. Shahbaz Mushtaq contributed 5% for editing and provided the comments on the manuscript.

Paper 2:

Van Thanh Tran, Duc-Anh An-Vo, Shahbaz Mushtaq, and Geoff Cockfield, 2022. "Nuanced assessment of livelihood resilience through the intersectional lens of gender and ethnicity: Evidence from small-scale farming communities in the upland regions of Vietnam". *Journal of Rural Studies, Vol. 92, pp. 68-78*. https://doi.org/10.1016/j.jrurstud.2022.03.011.

The overall contribution of Van Thanh Tran was 60% to the concept development, formal analysis, methodology, investigation, data curation,

drafting and revising the final submission; Duc-Anh An-Vo contributed 25% for concept development, formal analysis, editing and provided the comments on the manuscript. Geoff Cockfield contributed 10% for concept development, editing and provided the comments on the manuscript. Shahbaz Mushtaq contributed 5% for editing and provided the comments on the manuscript.

Paper 3:

Van Thanh Tran, Duc-Anh An-Vo, Geoff Cockfield, and Shahbaz Mushtaq (2022). "Farmers' perceptions and adaptation practices to climate variability risks, and their associated determinants: Evidence from small-scale farming communities in the upland regions of Vietnam". (Submitted to Weather and Climate Extremes).

The overall contribution of Van Thanh Tran was 60% to the concept development, formal analysis, methodology, investigation, data curation, drafting and revising the final submission; Duc-Anh An-Vo and Geoff Cockfield contributed 20 and 15% each respectively, for concept development, analysis, editing and provided the comments on the manuscript. Shahbaz Mushtaq contributed 5% each respectively for editing and provided the comments on the manuscript.

ACKNOWLEDGEMENTS

I would not have been possible to accomplish this thesis without the help and support of many kind people around me.

Foremost, I am very grateful to the Vietnamese Government and University of Southern Queensland (USQ) for providing a full scholarship to pursue my Ph.D. degree in Australia. I am deeply thankful to my supervisors, Prof. Geoff Cockfield, Dr. Duc-Anh An-Vo, and Prof. Shahbaz Mushtaq for allowing me to carry out my study under their supervision. Their understanding, patient guidance, advice, and intensive argument throughout my Ph.D. period contribute significantly to the success of my work. I also acknowledge the funding received from the German Federal Ministry for the Environment, Nature Conservation, Building and Nuclear Safety (BMUB).

I am grateful to all my colleagues at the Centre of Applied Climate Sciences: Kathryn Reardon-Smith, Thong Nguyen-Huy, Louis Kouadio, Jarrod Kath, and Thanh Mai, for their sharing and supports in various ways, not only in Ph.D. work discussions but also in social activities. Furthermore, I wish to extend my gratitude to fellow Ph.D. student at USQ, Vahid Rafiei for his encouragement and support. I am also thankful to my wonderful fieldwork assistants (Huyen Thi Duong, Tuyen Van Pham), interviewees, local officials, and organizations for their assistance and facilitations during the field surveys in Phu Yen district, Son La province.

I am greatly indebted to all USQ staff especially Christopher Dorney from Research Graduate School for his helpful support. He was always there to give advice on dealing with my personal and administrative problems throughout this journey. I wish to thank Barbara Harmes from USQ Student Services who was been constantly available for amazing proof reading of my writing whenever needed. I really appreciate the support from my colleagues, especially Dr. Yen Thi Bich Nguyen, Assoc. Prof An The Ngo at Viet Nam National University of Agriculture, Hanoi, Vietnam during my study period.

Keeping the most important to the last, I own my sincerest thanks to my lovely family. My parents, my brothers and my sisters have provided me all the love and unconditional support during this Ph.D. journey. To my children, Huong Giang Nguyen Tran, and Minh Duc Nguyen, you are my important energy source that motivate me complete this challenging journey. Your voice and your laugh make my long study trip meaningful, enjoyable, and unforgettable. To my wonderful soulmate-friend husband, Thuan Dat Nguyen, without you I would never made it. His inspiration and incredible support have made my long learning journey exciting and fulfilling. I am heartly thank for his accompany during my most stressful period to achieve my goals.

This research has been supported by an Australian Government Research Training Program Scholarship.

TABLE OF CONTENTS

ABSTRACT .	i
CERTIFICAT	ION OF THESISii
STATEMENT	OF CONTRIBUTIONiii
ACKNOWLEI	DGMENTS v
TABLE OF C	ONTENTSvii
LIST OF FIG	URESix
ABBREVIAT	IONS x
CHAPTER 1:	INTRODUCTION1
1.1.	Background and Problem Statement1
1.2.	Research Objectives5
1.3.	Scope of Research 6
1.4.	Thesis Organisation6
CHAPTER 2:	LITERATURE REVIEW9
2.1.	Vulnerability10
2.2.	Resilience12
2.3.	Adaptative Capacity14
2.4.	Assessing vulnerabiltiy and resilience16
CHAPTER 3 MINORITY E THE NORTH	: PAPER 1 - ASSESSING LIVELIHOOD VULNERABILITY OF THNIC GROUPS TO CLIMATE CHANGE: A CASE STUDY FROM WEST MOUNTAINOUS REGIONS OF VIETNAM
CHAPTER 4 RESILIENCE ETHNICITY: THE UPLANE	4: PAPER 2 - NUANCED ASSESSMENT OF LIVELIHOOD E THROUGH THE INTERSECTIONAL LENS OF GENDER AND EVIDENCE FROM SMALL-SCALE FARMING COMMUNITIES IN D REGIONS OF VIETNAM

CHAPTER 5	PAPER 3 - FARMERS' PERCEPTIONS AND ADAPTATION
PRACTICES	TO CLIMATE VARIABILITY RISKS AND THEIR ASSOCIATED
DETERMINA	NTS: EVIDENCE FROM SMALL-SCALE FARMING COMMUNITIES
IN THE UPLA	ND REGIONS OF VIETNAM56
CHAPTER 6:	CONCLUSIONS95
6.1.	Sumarry of important findings95
6.2.	Significance and scientific contribution of the study96
6.3.	Recommendations97
6.3.1.	Policy recommendations97
6.3.2.	Recommendations for future research98
REFERENCES	S 100

LIST OF FIGURES

Fig. 1.1. Thesis structure
Fig. 2.1. Conceptual framework between vulnerability, resilience, and adaptive capacity (<i>Adapted from Cutter et al. (2008)</i>)
Fig. 2.2. The basic understanding of adaptive capacity's role in influencing vulnerability. Adaptive capacity influences a system's vulnerability by moderating exposure and sensitivity (<i>Adapted from Engle (2011)</i>)
Fig. 2.3. A primary depiction of adaptive capacity's role in managing resilience (<i>Adapted from Engle (2011)</i>)16

ABBREVIATIONS

Abbreviations	Definition
NMRs	Northwest Mountainous Regions of Vietnam
USQ	The University of Southern Queensland
HMDC	Hydro-Meteorological Data Centre of Vietnam
MONRE	Ministry of Natural Resources and Environment
GoV	The Government of Vietnam
NTP	National Target Program for Climate Change
IMHEN	Vietnam Institution of Meteorology, Hydrology and
	Climate Change
IPCC	Intergovernmental Panel on Climate Change
SLA	Sustainable Livelihood Approach
LVI	Livelihood Vulnerability Index
LVI-IPCC	Livelihood Vulnerability Index by IPCC Framework
	Approach
MSD	Mean standard deviation
HLRA	Household Livelihood Resilience Approach
LRI	Livelihood Resilience Index
(KIIs)	Key Informant Interviews

CHAPTER 1: INTRODUCTION

1.1 Background and problem statement

Global concerns about climate change are increasing. According to projections issued by the IPCC in 2007, the world's climate will continue to change at unprecedented speeds. Increases in temperature, changes in precipitation patterns, and an increase in the frequency of extreme weather events such as floods and droughts are well-documented (IPCC 2007; IPCC 2012). These have the potential to threaten human welfare and security, and consequently human life quality, in profound and unprecedented ways. In fact, climate change is already having an adverse impact on nature and humans worldwide. Basically, climate change and its stressors affect humans by destabilising livelihoods, especially for poor households (Tanner et al. 2014). Climate change has impacted livelihoods, leading to an increasing vulnerability, reducing the possibilities of securing livelihoods and eradicating poverty. Variability in climate affects economies worldwide, especially in low- and middle-income countries owing to the heavy dependency on nature-sensitive sectors like agriculture and forestry. In addition, many low- and middle-income countries are experiencing higher risks from unusual changes in climate phenomena compared to high income countries because of both their location and geography (Maharjan and Joshi 2013). Social-ecological systems are under intense pressure from climate change and variability (Berkes and Jolly 2001; Adger and Barnett 2009; Bardsley and Sweeney 2010). Climate change has been and will continue to lead to comprehensive and profound changes in global development and security, particularly in water, energy, food, economy, employment, culture, and trade (IPCC 2014).

Vulnerability and low resilience have been identified as particular concerns for developing countries such as Vietnam. Vietnam is frequently and severely impacted by a changing climate through sea level rise, storms, droughts, flash floods, and landslides (Dasgupta et al. 2009; Few & Tran 2010; Eckstein et. al 2017). According to the Ministry of Natural Resources and Environment (MONRE 2016), there has been an increase of 0.26°C per decade in annual average temperature since the 1970s, meanwhile, yearly rainfall has varied across the country with a declining trend over the Northern regions but an increasing trend over the Southern areas. Over the past two decades, climate hazards have contributed to more than 13,000 deaths and average yearly asset losses of more than US\$6.4 billion (equivalent to 1.5% of Vietnam's GDP) (Bank 2017; MONRE 2017).

Unfavourable changes in climate are expected to have a wide range of effects on various sectors, particularly in agriculture because this sector is highly exposed to climatic conditions (Mestre-Sanchís & Feijóo-Bello 2009; IPCC 2013; Houser et al. 2015). Changing weather patterns have caused a significant change in rice production (Krishnan et al. 2007; Babel et al. 2011; Shrestha et al. 2013; Burney & Ramanathan 2014), yield reductions from 2% to 26% in Southeast Asia countries (Weiss 2009; Zhai & Zhuang 2012), and yield declines from 5% to 15% in a variety of staple food crops in Africa (Roudier et al. 2011; Blanc 2012; Salack et al. 2015). The natural relationships between climatic factors and the agricultural sector have been thoroughly researched (Welch et al. 2010; Desch et al. 2012; Fisher et al. 2012; Lobell et al. 2013; Roberts et al. 2013; Burke & Emerick 2016; Chen, Chen & Xu 2016). These studies have considered factors such as water resources, agriculture, fisheries, forestry, human health, infrastructures, and ecological systems, which are considered essential to livelihoods (Kohler & Maselli 2012). Poverty, rural population, and dependency on agriculture are also other factors enhancing the level of vulnerability (Asafu-Adjaye 2014).

Vietnam is highly reliant on natural resources compared to other middle-income nations in Southeast Asia (World Bank, 2010). Furthermore, the people's main livelihood is from the agricultural sector. However, agriculture's contribution to GDP has been declining over the last two decades. Agriculture contributed 38% of GDP in 1990 and employed 73% of the labour force. By 2015, agriculture's contribution to GDP had dropped to 18%, but with about 44% of the workforce employed in this industry. Therefore, climate change and its induced risks still contribute to significant sectoral and household insecurity. For instance, due to salinity intrusion and severe droughts in Mekong River Delta, rice production significantly decreased and caused losses of approximately 15,000 billion VND (about US\$646 million) to the national economy. As a consequence, many farmers were forced to abandon their farms and seek employment opportunities in cities or surrounding urban areas.

Poor and marginalised groups in developing countries, such as Vietnam are most vulnerable to climate change because of the high level of poverty, lack of social safety systems, weak adaptive capacity, difficult accessibility to education programs, health care services and alternative means of production, low education, large family structure and size (Watson et al. 1996; Moser 1998; IPCC 2007; Lemos et al. 2007; Skoufias et al. 2011; Kurukulasuriya & Rosenthal 2013; Shah et al. 2013; Masud et al. 2014; Alam et al. 2018). Climate change is expected to disproportionately affect the world's poorest and most vulnerable people, who mostly live in rural areas, by adversely affecting water availability and supply, infrastructure, agricultural incomes, and food security (Beaumont et al. 2011; Lynn et al. 2011; IPCC 2014; Masud et al. 2014).

Within the agricultural sector, there may however be sub-groups with particularly high levels of vulnerability and low levels of resilience. One factor in this may be minority ethnicity, which could indicate multiple disadvantages in relation to access to resources, government influence and programs, education and so on. Vietnam has 54 recognized ethnic groups with varying lifestyles, cultures, and languages, of which 53 groups are classified as minorities. Ethnic minority groups account for more than 50% of the total poor in Vietnam, despite comprising less than 15% of the population (based on the Vietnam Household Living Standard Survey 2014). Generally, these minorities often reside in less productive, geographically remote, or mountainous areas with poor access to infrastructure, lack of health and educational facilities, and non-farm opportunities. They have higher poverty rates than majority groups (Imai et al. 2011; Tran et al. 2015). People living in such situations also often experience substantial food shortage and low quality water due to impacts of climate hazards such as drought, floods, and storms.

Located in the North of Vietnam, the Northwest Mountainous Regions (NMRs) are highly sensitive to slight changes in the frequency and extent of natural disasters because of comparatively fragile ecosystems, unstable geology, and complex topography (UNDP 2015). In addition, these regions are also ranked the poorest and highest inequality regions of Vietnam, with 95.6% of the population being ethnic minorities (World Bank 2010; GSO 2015; Tuyen 2016). Within the study region, the level of education, especially, among ethnic minorities, is far below the national average (Ha 2007). The NMRs are hilly, remote regions without advanced infrastructure, leading to significant barriers to access to even close cities or towns for living activities such as shopping, attending schools or seeking medical assistance or services.

The Government of Vietnam (GoV) is increasingly recognizing threats from climate change, and a National Target Program for Climate Change (NTP) was adopted in December 2008. The Program document, however, has little to say about how adaptation will take place and who will be the most vulnerable populations beyond noting mountainous areas are presumed to be amongst the most vulnerable places, with sustainable agricultural productivity and livelihoods the most serious challenges (GoV, 2008). Recently, the Vietnam Institution of Meteorology, Hydrology and Climate Change (IMHEN & UNDP, 2015) also notes that, in Vietnam, vulnerability is concentrated in poor communities, and it is crucial to address the underlying causes of vulnerability in the context of climate change to achieve sustainable development goals. Given that fact, a study of climate change impacts on the livelihoods of mountainous farmers is particularly critical.

In this regard, a number of studies have been conducted to better understand the impacts of climate change on agriculture (Do 2013; Anh 2016; Benson 2017), to assess rural households' vulnerability (CARE 2013; Nguyen, Vo & Chu 2013; Huynh & Stringer 2018; Nguyen & Leisz 2021), and to investigate the factors that influence farmers' decisions to adapt to climate change (Nam 2011; Le Dang et al. 2014; Van et al. 2015; Trinh et al. 2018). However, almost all these studies focus on the two Deltas (the Red River Delta and the Mekong River Delta) and the central area. In contrast, the Northwest Mountainous Regions of Vietnam, which are particularly vulnerable to climate change, have not yet drawn much of researchers' attention, particularly in relation to ethnicity. For the abovementioned reasons, there is a high demand for research that explores the vulnerability levels of different ethnic groups in mountainous regions, how they perceive and response to climate change and climate hazards.

Against this background, the study of "Vulnerability, Resilience and Adaptation to Climate Change in the Northwest Mountainous Regions of Vietnam" is conducted with the expectation for filling a fundamental knowledge gap and adding further information and insights in the existing literature in explaining the impacts of climate change on mountainous households' vulnerability, their resilience, and their decision-making behaviours. Hence, the findings of the current work will be useful for designing appropriate policy practices in order to enhance farmers' capacity and resilience toward future climate hazards not only in Vietnam but also in other countries having similar economics, social and graphical contexts.

1.2 Research Objectives

The general objective of this study is to gain a comprehensive picture of ethnic households' livelihoods, to explore the vulnerability level of farm households and their perceived resilience, and to investigate key factors influencing adaptation choices to climate change among ethnic groups in the Northwest Mountainous Regions of Vietnam.

The overall objective is divided into the following three sub-objectives:

- To assess livelihood vulnerability among groups and reveal the factors affecting their vulnerability to climate change.

- To explore differences in households' perceived livelihood resilience among members of minority groups and genders.

- To identify how farmers have adapted to climate change, and to determine major factors influencing farmers' adaptation choices.

1.3 Scope of Research

To achieve these objectives, the work was segmented into distinct tasks as follows:

- The lists of livelihood vulnerability/resilience indicators were selected from the literature and then reviewed by local experts. Questionnaires were created, pre-tested with local people, and revised before formal interviews. Household surveys were conducted by the student and one local assistant.

- The level of household livelihood vulnerability was assessed and analysed based on a livelihood vulnerability index.

- A nuanced intersectionality analysis was conducted based on household livelihood resilience approach to understand varying perceptions of livelihood resilience among groups and genders.

- Using survey data, households' perceptions of climate variables were reviewed. We also investigated adaptation practices adopted across the studied communities and their drivers using logit regression modelling.

1.4 Thesis Organisation

The study is organized into six chapters. Chapter 1 presents the research background and problem statement, research objectives, and scope of research. Chapter 2 provides the literature review the concepts of vulnerability, resilience and adaptive capacity used for the current study. In

Chapter 3, the community vulnerability level and factors influencing households' vulnerability under the impacts of climate change are analysed. This chapter was published in the journal Sustainability as a peer-review paper entitled "Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam". Chapter 4 explores how perceptions of livelihood resilience differ by gender and groups and identifies the main factors contributing to better livelihood resilience. This chapter was published in the Journal of Rural Studies as a peer-review paper entitled "Nuanced assessment of livelihood resilience through the intersectional lens of gender and ethnicity: Evidence from small-scale farming communities in the upland regions of Vietnam". Chapter 5 addresses the question of how farmers have responded to climate change for controlling its adverse impacts. At the time of submission, this chapter had been submitted to Weather and Climate Extremes as a peer-review article entitled "Farmers' perceptions and adaptation practices to climate variability risks and their associated determinants: Evidence from small-scale farming communities in the upland regions of Vietnam". Chapter 6 provides a general conclusion drawn from the previous chapters, expected significant contributions of the study, and some recommendations for future research.



Fig. 1.1. Thesis structure.

CHAPTER 2: LITERATURE REVIEW

There are different understandings and definitions of, and linkages between, the concepts of vulnerability, resilience, and adaptive capacity. Birkmann (2006), Folke (2006), and Adger (2006) considered resilience as an integral part of adaptive capacity (Fig. 2.1A); meanwhile, Burton et al. (2002) and O'Brien (2004a) viewed adaptive capacity as a key component of vulnerability (Fig. 2.1B). In the context of climate change, vulnerability is defined as a function of three main factors: exposure, sensitivity, and adaptive capacity (Adger 2006). A third view sees them as nested concepts within an overall vulnerability structure (Fig. 2.1C) (Turner et al. 2003a; Gallopín 2006). In natural hazard research, resilience is understood as the ability to survive and cope with a disaster with minimum damage (Cutter et al. 2008). Resilience is also identified as an outcome in relation to the ability to cope with a hazard event and is embedded within vulnerability (Fig. 2.1D) (Manyena 2006). In global perspectives, scientists often embed adaptive capacity within resilience (Fig. 2.1E) (Bruneau et al. 2003; Tierney & Bruneau 2007; Paton & Johnston 2017).



Fig. 2.1. Conceptual framework between vulnerability, resilience, and adaptive capacity (*Adapted from Cutter et al. (2008)*).

2.1 Vulnerability

The concept of vulnerability has been used in different fields of research, for example, natural hazards, food security, and political ecology (Kelly & Adger 2000; McLaughlin & Dietz 2008), yet there is little agreement on its meaning. Some researchers consider vulnerability according to factors such as exposure and sensitivity to perturbations or external stresses and the capacity to adapt (O'Brien 2004a). Other researchers look at vulnerability in relation to exposure, sensitivity, and resilience (Turner et al. 2003a, b). In other cases (Ford & Smit 2004; Smit & Wandel 2006), vulnerability is viewed as a function of exposure and adaptive capacity following the argument that exposure and sensitivity cannot be separated, and resilience is a subset of adaptive capacity (Gallopín 2006). In this conceptualisation, vulnerability, resilience, and adaptive capacity are seen as nested concepts within an overall vulnerability structure (Fig. 2.1C). However, both capacity to cope or respond and adaptive capacity are considered components of a system's resilience (Turner et al. 2003a, b). Adger (2006) reviewed existing approaches to vulnerability from social and natural sciences and concluded that three key components most commonly conceptualise vulnerability: exposure to perturbations, sensitivity to perturbation, and adaptive capacity.

In terms of climate studies, the IPCC third assessment report (IPCC 2001: 6) describes vulnerability as "*the degree to which a system is susceptible to, or unable to cope with, the adverse effects of climate change, including climate variability and extremes*". Additionally, in the IPCC Fourth Assessment Report, Cardona et al. (2012) describes vulnerability as a function of exposure, sensitivity, and adaptive capacity, as also reflected by, for instance, Brooks (2003), O'Brien (2004a), Füssel and Klein (2006), Füssel (2007), and O'Brien et al. (2008). However, Eakin and Luers (2006) claim that the above components and their relationships are not well-defined and considered vulnerability as a dynamic property of a system in which humans are constantly interacting with the biophysical

environment. It is an important task to clarify the meaning and application of the vulnerability concept in these analyses (Renaud et al. 2010). Ford and Smit (2004) conceptualised vulnerability to climate change as a function of exposure and adaptive capacity.

Exposure here reflects the susceptibility of people and communities to climatic conditions, and adaptive capacity reflects the ability or potentiality of a community to address, plan for, or adapt to exposure. In this conceptualisation, vulnerability at a local level is viewed as being conditioned by social, economic, cultural, political, and climatic conditions/processes, operating at multiple scales over time and space, which affect community exposure and adaptive capacity (Ford et al. 2008). Ford et al. (2008) argued this is broadly consistent with other approaches to vulnerability, including Turner et al. (2003a), Smit and Wandel (2006), and Eriksen, Brown and Kelly (2005). Furthermore, Füssel (2007) argued that there is no single 'correct' or 'best' conceptualization of vulnerability that would fit all assessment contexts.

A localised concept of vulnerability (Turner et al. 2003a, b; Ford & Smit 2004; Smit & Wandel 2006) is applied in this study because vulnerability and adaptation to climate change are exceedingly variable and linked to local contexts (Leichenko & O'brien 2002; O'Brien et al. 2004b). Therefore, one-size-fits-all approaches are likely to miss socio-economic and political-institutional dynamics of vulnerability and, hence, risk any assessment being ineffective, if not counterproductive (Tschakert 2007). Vulnerability at a local level is identified as being conditioned by social, economic, cultural, political, and climatic conditions/processes, operating at multiple scales over time and space, which affect community exposure, sensitivity, and adaptive capacity (Ford & Smit 2004; Ford et al. 2008). This study will assess the vulnerability at community level by the composite livelihood index (LVI) and the livelihood index (LVI-IPCC) based on IPPC's three contributing factors: exposure, sensitivity, and adaptive capacity.

2.2 Resilience

Resilience is a perspective for understanding how co-evolving societies and natural systems can cope with and develop from disturbances and change (Manyena 2006). The concept of resilience is mainly used in the study of ecosystem dynamics (Holling 1973); however, it has also been applied to social (Adger 2000) and social-ecological systems (Berkes & Folke 1998). The term resilience is often used as synonymous with the notion of 'bouncing back', from its Latin root, meaning "to jump back" (Paton 2006: 7). It implies a capability to return to a previous state. However, the concept does not necessarily capture the reality of social/socio-ecological systems experiencing perturbations. For example, changes to the physical, social, and psychological reality of social life following a disaster can present community members with a new reality which may differ in several ways from conditions pre-disaster. It is the changed reality to which people must adapt (Paton 2006).

In ecological systems, Holling (1973: 9) first defined resilience as "the capacity of a system to absorb and utilize or even benefit from perturbations and changes, and so persist without a qualitative change in the system's structure." The concept of ecological resilience was added to vulnerability discussions. It has contributed to a productive exchange of ideas on assessing and understanding vulnerability concerning global environmental change and various stresses and shocks acting on and within coupled human-environment systems. The concept of resilience has recently begun to be more widely considered in the social sciences (Folke 2006). Resilience is defined as communities' ability to withstand disturbances to maintain their social infrastructures (Adger 2000). However, resilience is a cumbersome concept for the social sciences, at least when trying to speak of resilience at the systems level (Davidson 2010; Duit et al. 2010). Duit et al. (2010) argued that the concept of social-ecological systems resilience should be used to avoid confusion.

A definition of resilience incorporating social-ecological linkages has been developed (Carpenter et al. 2001; Berkes et al. 2003). This definition includes consideration of the amount of change a system can undergo and retain essentially the same function, structure, and identity; the degree to which the system is capable of self-organisation; and the degree to which the system represents the capacity for learning and adaptation (Folke et al. 2002). Carpenter et al. (2001: 765) defines resilience as "the magnitude of disturbance that can be tolerated before a social-ecological system moves to a different state controlled by a different set of processes". A 'socialecological system' in this sense encapsulates ecosystems and their human use by communities and institutions. For Berkes, Colding and Folke (2003), the resilience of social-ecological systems is related to sustainability.

Sustainability is defined as the adaptive capacity to deal with change and the maintenance of ecological systems' capacity to support associated social and economic systems. For Folke (2006), the concept of socialecological resilience, meanwhile, involves the intersection of factors such as adaptive capacity, transformability, learning, and innovation. Resilience is taken to mean the capacity to deal with change and continue developing; hence, resilient communities will be those with coping abilities to withstand, recover from stresses/shocks and surprises, and, specifically, adapt to climate variability and change. They also can turn shocks/surprises into opportunities through learning. This concept of resilience is consistent with that of many others who define resilience as the ability of a system to a) absorb shocks and retain its primary function; b) self-organise; and c) innovate and learn in the face of disturbances (Carpenter et al. 2001; Folke 2006). However, Brown (2013) suggests there was not enough attention to social-ecological systems' social or political side within resilience thinking and research. A livelihood perspective concept was developed to address this criticism. For example, livelihood resilience is viewed as the accumulation of people's capacity through generations to maintain and improve their livelihood opportunities and well-being in the case of environmental, economic, social, and political instabilities (Tanner et al. 2014). Therefore, it is necessary to place people at the centre of resiliencebuilding analysis and to emphasize the role of human organisation, rights, and capacity in preparedness and adaptation to shocks. Resilience-building programs or interventions also need to address the question of "resilience for whom?" to establish objectives (Brown 2013). In the light of this development, Constas (2014) emphasised resilience capacity as a way of looking at measurable resources and assets that may help individuals, households, or communities prepared for and respond to shocks. Another study quantified household resilience by using people's own perceptions called "subjective resilience" (Jones & Tanner 2015). Following the current focus of livelihood in both academic research and resilience-building projects, we draw from the idea of Jones and Tanner (2015) to assess the subjective livelihood resilience at the household level in the research area.

2.3 Adaptive capacity

Adaptive capacity is identified as the ability to adapt, and adaptive capacity is primarily conceptualized separately to either a vulnerability or a resilience framework. The IPCC defines adaptive capacity as "the ability of systems, institutions, humans, and other organisation to adjust to potential damage, take advantage of chances and respond to consequences" (Smit & Pilifosova 2003; Adger et al. 2007; IPCC 2013: 549). Adaptive capacity affects vulnerability (Section 2.1) by modulating exposure and sensitivity (Adger et al. 2007) or by addressing and reducing the vulnerability of a system under human actions affecting the system's biophysical and social factors (Eakin & Luers 2006). Figure 2.2 represents a conceptual view of adaptive capacity concerning vulnerability through moderating exposure and sensitivity. While there is an ongoing argument about definitions and boundaries between exposure, sensitivity, and adaptive capacity (Gallopín 2006; Füssel 2007), adaptive capacity is generally considered a fundamental role in a positive attribute of a system for reducing vulnerability.



Fig. 2.2. The basic understanding of adaptive capacity's role in influencing vulnerability. Adaptive capacity influences a system's vulnerability by moderating exposure and sensitivity (*Adapted from Engle (2011)*).

In resilience literature, adaptive capacity often refers to 'adaptability', or the capacity of actors in the system to influence resilience; in socialecological systems this amounts to humans' capacity to manage resilience (Walker et al. 2004). Human actions affect resilience through interactions between a system's human and environmental elements (Walker et al. 2006). Folke (2006) considered adaptive capacity as a property that can support people in the creation of a new system state when the existing state is somehow untenable. Adaptive capacity affects a social-ecological system by moderating between maintenance of the current situation and movement of the system to a new more 'desirable' state, where the concept of 'desirability' is based on social factors (Robards et al. 2011). Depending on how 'desirability' is negotiated/defined within a given system (i.e., system-maintaining or system-altering), more adaptive capacity increases the likelihood of achieving a desirable state (Fig. 2.2).

Resilience approaches point to governance and institutions as the two main variables affecting a socio-ecological system's adaptive capacity. For example, Lebel et al. (2006) recommend that enhancing a society's capacity to manage for resilience is essential for sustainable developmentrecognising that resilience enhancement might be more desirable in some situations, while transformation might be more desirable in others. Systems with higher adaptive capacity are likely to 'end up' in a desirable system state, while systems with lower adaptive capacity are likely to move to a less desirable system state (Fig. 2.3).



Fig. 2.3. A primary depiction of adaptive capacity's role in managing resilience (*Adapted from Engle (2011)*).

2.4 Assessing vulnerability and resilience

Socio-economic, bio-physical, and integrated assessment approaches are the three major conceptual approaches applied to analyse vulnerability to climate change (Deressa et al. 2008). Of these, the third approach combines both socio-economic and bio-physical approaches to define vulnerability. Vulnerability mapping (O'Brien et al. 2004b; Kumar & Tholkappian 2006) is a good example of this approach. Both socio-economic bio-physical factors are systematically combined to and identify vulnerability. The econometric method originated from the poverty and development literature and utilizes household-level socio-economic survey data; therefore, it can assess the level of vulnerability of different social groups (Deressa et al. 2008). The indicator method firstly selects numerous leading indicators from the whole set of potential indicators and then systematically combines these indicators chosen to estimate the levels of vulnerability (Adger & Kelly 1999; Kumar & Tholkappian 2006; Deressa et al. 2008; Moreno & Becken 2009; Nyong et al. 2009; Nelson et al. 2010; Seidl et al. 2011; Maiti et al. 2015). In the present study, the livelihood vulnerability index (LVI and LVI-IPCC) is used to understand and assess the vulnerability at community level (details in Chapter 3). Furthermore, subjective livelihood resilience at household level was measured by applying the Household Livelihood Resilience Approach (HLRA) (Quandt 2018) (details in Chapter 4). This approach uses various resilience indicators adopting from the Sustainable Livelihoods Framework (Chambers & Conway 1992), including five types of livelihood capital assets: financial, human, social, physical, and natural capital.

This chapter provided a review of the concepts of vulnerability, resilience and adaptive capacity used for the current study. Also, through literature review, our research shows the expectation for filling a fundamental knowledge gap in the field. Assessing the level of vulnerability in various minority ethnic communities in the Northwest Mountainous Regions of Vietnam can be used to address such a gap, considering particularly ethnicity (Chapter 3).

CHAPTER 3: PAPER 1 - ASSESSING LIVELIHOOD VULNERABILITY OF MINORITY ETHNIC GROUPS TO CLIMATE CHANGE: A CASE STUDY FROM THE NORTHWEST MOUNTAINOUS REGIONS OF VIETNAM

Originally published as: VT Tran, D-A An-Vo, S Mushtaq, G Cockfield, 2021. Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam. Sustainability 2021, 13, 7106. https://doi.org/10.3390/su13137106.



Article

Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam

Van Thanh Tran 1,2,*, Duc-Anh An-Vo 1, Geoff Cockfield 3 and Shahbaz Mushtaq 10

¹ Centre for Applied Climate Sciences, University of Southern Queensland, Toowoomba 4350, Australia;

MDPI

- Duc-Anh.An-Vo@usq.edu.au (D.-A.A.-V.); Shahbaz.Mushtaq@usq.edu.au (S.M.)
- ² Faculty of Natural Resources and Environment, Vietnam National University of Agriculture (VNUA), Trauquy, Gialam, Hanoi 12406, Vietnam
- ³ Centre for Sustainable Agricultural Systems, University of Southern Queensland,
- Toowoomba 4350, Australia; Geoff.Cockfield@usq.edu.au
- * Correspondence: VanThanh.Tran@usq.edu.au or tranthanhvan@vnua.edu.vn

Abstract: Climate variability, climate change, and extreme events can compound the vulnerability of people heavily reliant on agriculture. Those with intersecting disadvantages, such as women, the poor, and ethnic minority groups, may be particularly affected. Understanding and assessing diverse vulnerabilities, especially those related to ethnicity, are therefore potentially important to the development of policies and programs aimed at enabling adaptation in such groups. This study uses a livelihood vulnerability index (LVI) method, along with qualitative data analysis, to compare the vulnerability of different smallholder farmers in Son La province, one of the poorest provinces in Vietnam. Data were collected from 240 households, representing four minority ethnic groups. The results indicated that household vulnerability is influenced by factors such as income diversity, debt, organizational membership, support from and awareness by local authorities, access to health services, water resources, and location. Results revealed that two of the ethnic groups' households were, on average, more vulnerable, particularly regarding livelihood strategies, health, water, housing and productive land, and social network items when compared to the other two ethnic groups. The study shows the need for targeted interventions to reduce the vulnerability of these and similarly placed small ethnic communities.

Keywords: livelihood vulnerability; agricultural dependency; climate change

1. Introduction

Vietnam is ranked as one of the world's ten most vulnerable countries to climate change and climate events such as rising sea levels, storms, floods, and droughts [1–3]. Under increasing climate variability, people whose livelihoods rely mostly on agricultural activities are relatively vulnerable, particularly in developing countries [4]. Reference [5] indicates that coastal people in Vietnam have generally higher vulnerability to climate change because nearly 60% of livelihoods are based on aquaculture and agriculture, whereas the mountainous regions have unstable and complex topography with poorer economic prospects, and people in those regions are highly sensitive to slight changes in the frequency and severity of climate events [6,7]. Vulnerability and adaptation research in Vietnam has largely focused on coastal areas, especially in the Mekong River Delta, with most work focusing on assessing the direct impacts of climate change. Significant threats include increases in the frequency and intensity of droughts and sea level rise driving saline intrusion in the Mekong River Delta, causing the loss of land for rice production, which could threaten national food security [8]. There may, however, be differences in the degree of vulnerability and capacity to adapt amongst different groups, especially considering

check for updates

Citation: Tran, V.T.; An-Vo, D.-A.; Cockfield, G.; Mushtaq, S. Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam. *Sustainability* 2021, 13, 7106. https://doi.org/10.3390/su13137106

Academic Editor: Hossein Azadi

Received: 13 May 2021 Accepted: 21 June 2021 Published: 24 June 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.

۲ (cc)

Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

https://www.mdpi.com/journal/sustainability

various forms of social disadvantage. To develop proper adaptation strategies and solutions/policies to reduce rural households' vulnerability and to improve their resilience, it is very important to understand the livelihood vulnerability of rural households, especially in countries depending heavily on agriculture.

Livelihood vulnerability can be a function of both physiological and social factors [9]. Physiological vulnerability is the extent to which communities are exposed to physical effects such as sea-level rise and an increase in sea temperature, and/or atmospheric temperature. Such exposure to climate change increases rural livelihood vulnerability and reduces households' ability to cope with climate risks, shocks, and stress [10]. Rural households often have limited assets and thus adaptive capacity [11]. The social vulnerability can include factors such as relative inequality, the degree of urbanisation, and the rate of economic growth [9].

Vulnerability assessments have become a core means of understanding development challenges and climate change influence in many contexts. Such assessments can encompass the numerous methods utilized to systematically consider interactions between humans and their environmental surroundings, including physical and social aspects [12]. Approaches to vulnerability assessment include historical narrative, comparative analysis, statistical analysis, indicator-based methods, and agent-based modelling. Recently, the indicator-based method has been widely used to assess vulnerability to climate change and climate-induced disasters [13,14]. Almost all the approaches use indicators to characterize and quantify the different dimensions of vulnerability, with the common practice being to combine the diverse indicators into a single composite index [12,15]. The indicator approach has been used at different scales and domains to quantify system dynamics [13,16-18]). The Sustainable Livelihood Approach (SLA) (Figure 1) has been used to understand household livelihoods and to plan community development programs. This approach considers five types of household's assets i.e., natural, social, financial, physical, and human, and uses multiple indicators to assess exposure level to natural disasters and climate change. Households' economic characteristics affect households' adaptive capacity, and the characteristics of health, food and water resources determine the household's sensitivity to climate change impacts [19]. A major work in livelihood vulnerability assessment is that of [12], who developed two approaches. They first expressed LVI as a composite index, comprising seven major components. The second approach was based on the vulnerability definition of the Intergovernmental Panel on Climate Change (IPCC), whereby they decomposed the seven components into three: based on exposure, sensitivity, and adaptive capacity. The LVI approaches consist of variables indicating the level of exposure, sensitivity, and adaptive capacity to climate-induced disasters (for example, droughts and floods, landslide, etc.) and climate change. The LVI indicates a way to understand how vulnerability varies across time and space and to identify the main factors contributing to vulnerability, highlight strategies reducing the vulnerable level, and also evaluate how efficient these strategies are in different social and ecological environments [15]. In the past decade, the LVI has become a means of assessing farmers' vulnerability to climate change and disasters around the world [4,15,20-23].



H = Human capital S = Social capital P = Physical capital F = Financial capital N = Natural Capital

Figure 1. The asset pentagon lies at the core of sustainable livelihood approach, within the vulnerability context [24].

The Northwest mountainous region (NMR) of Vietnam is highly sensitive to slight changes in the frequency and extent of natural disasters with its fragile ecosystems, unstable geology, and complex topography [7]. The NMR is home to numerous marginalized ethnic minority groups that experience, in relative terms, extremely low-income levels and poor health care. The region is ranked the poorest and highest inequality region of Vietnam (with the overwhelming majority of the population (95.6%) being ethnic minorities) [25-27]. The level of education, especially among ethnic minorities, is far below the national average [28]. People living in this area also often experience substantial food shortage and low water quality due to climate change such as extreme weather events. Vietnam Institution of Meteorology, Hydrology and Climate Change [29] notes that, in Vietnam, vulnerability is concentrated in poor communities and it is crucial to address the underlying causes of vulnerability in the context of climate change to achieve sustainable development goals. Despite recognition of the need, there has been little attention focused on the vulnerability of communities' livelihood systems to climate change in the mountainous regions of Vietnam and specifically on the challenges faced by ethnic minority communities. Further, NMR is a hilly remote region without advanced infrastructure, leading to significant barriers to access to even close cities or towns for living activities such as shopping, attending schools or seeking medical assistance or services. For the above mentioned reasons, work focusing on the NMR is particularly necessary to develop appropriate strategies in support of reducing the poverty and vulnerability of rural households and ethnic minority groups. Importantly, previous work conducted by [23] to assess household livelihood vulnerability to climate change in the NMR did not focus on the ethnicity perspective, which remains a major gap.

This research aims to explore the livelihood vulnerability of different ethnic groups living in the Phu Yen district, Son La province in the NMR. The ethnic groups in the Phu Yen district were selected as typical of communities in the region. We consider major factors driving different assessed outcomes among the ethnic groups which could then be addressed in climate change adaptation policies and programs. We hypothesized, based on previous vulnerability studies and studies of poverty in Vietnam, that minority ethnic groups would have relatively high levels of vulnerability, generally, and therefore in relation to climate change.

We apply the LVI [12] and reference the work of [30] but have modified or added a number of new indicators relevant to the Son La province in the NRM to better understand the livelihoods of local minority people and explore the main factors affecting the vulnerability of households to climate change. This research contributes to the literature concerning the assessment of vulnerability of rural households and provides a reference for policy making aimed at helping people living in similar economic and natural regions. More specifically, this research assists in developing targeted policy interventions aiming at improving resilience of the marginal ethnic groups in mountainous regions of Vietnam.

2. Materials and Methods

2.1. Description of the Study Area

The study was carried out in the Phu Yen district, Son La province (Figure 2). Son La is considered highly vulnerable to climate change because of its topography and geography [31]. According to the Ministry of Labour, War Invalids and Social Affairs of Vietnam (Decision 1095/OD-LDTBXH dated 8 August 2016), Son La was the province having the third-highest number of poor households in Vietnam during 2016-2020. Phu Yen is the third-largest rice-producing district of the northwest region but also one of the five poorest districts of Son La province. Phu Yen district is receiving support from the 'National Target program for Sustainable Poverty Reduction in the 2016-2020 period'. There are 27 villages in Phu Yen district, of which 14 are located in the highlands and 11 belong to the groups considered as "especially difficult communes" specified under Program 135. The total natural area of Phu Yen district is 1227 km² with a population of approximately 116,000 people. Agricultural production includes intensive rice, fruits and crop. Phu Yen district has four main ethnic groups: Thai, Muong, Dao, and Hmong. Among them, Thai and Muong make up the majority of the population. Some groups may have several compounding disadvantages, including isolation, social and economic exclusion, and a very high dependence on agricultural production. The Hmong and Dao people, for example, often live in high mountainous areas far from district/commune centres while the remaining ethnic groups mainly live in lowland areas (valleys) and/or near the district centre (Table 1). Thus, they are further from health and education services and labour markets. The people in the study area often experience economic loss because of natural hazards such as droughts and hot winds in the dry season, flash floods and landslides in the rainy season, and cold spells and frosts in winter [32]. In summary, the study site has known disadvantaged groups and high exposure to climatic variability, events, and change.



Figure 2. Map showing the case study. (a) is the map of Son La province, Vietnam; (b) showing the study site.

Characteristics	Thai	Muong	Dao	Hmong
Average age of household head (age)	49.7 ± 8.44	49.6 ± 8.66	41.3 ± 7.68	41.6 ± 10.50
Average family members (number)	4.9 ± 1.24	4.7 ± 1.00	5.05 ± 0.97	7.0 ± 2.12
Main income source (agricultural income, %) With some non-agricultural income (%)	92.5 78.5	89.7 83.5	95.4 77.3	92.9 46.4
With outside community work (%)	61.3	60.8	45.4	42.9
Limited formal education (%)	54.8	54.6	59.1	67.9
Average distance to district centre (km)	3.36 ± 1.81	7.5 ± 5.33	15.2 ± 6.69	8.4 ± 1.85

Table 1. Brief characteristic for four ethnic groups in the Phu Yen District (source: from surveys).

2.2. Data Collection

The questionnaire, largely based on items used in previous studies but applied elsewhere [12,15,22,33], consisted of eight sections, including household demographic profile, livelihood strategies; social networks and finance; health; food; water supply; housing and productive land; natural disasters and climate variability. There were initial in-depth interviews with experts from organizations such as Statistical Departments, Agriculture Department, Meteorological Centre, and the People's Committee at both provincial and district level, in order to better understand the research context and to select study sites in Phu Yen district. A list of suggested components related to vulnerability assessment to climate variability and climate events was given to local officials and experts in the fields of agriculture and climate for advice on which components were relevant to the locality. These components were then revised for the household survey (see Appendix A). A survey of 240 households in the Phu Yen district was conducted from December 2018 to January 2019. Households were randomly selected from lists of all communities. As the primary purpose of this study is to focus on understanding and assessing ethnic minority vulnerability, all participants were categorised into one of four ethnic groups. Interviews were conducted by the lead author and one local assistant. Generally, interviews were conducted only with the head of the household but if he/she was not available, the main labourer was interviewed. Each interview took 1 to 1.5 h on average and was conducted in the Vietnamese language. The local people, including the various minority groups, have mostly used this language in daily communication. Surveys were conducted with the approval of the Human Research Ethics Committee of the University of Southern Queensland. Data were entered into, checked, and analysed within Excel software 2010. Secondary data on daily minimum and maximum temperatures, and daily precipitation were collected from the Phu Yen meteorological station and also obtained from the Hydro-Meteorological Data Centre of Vietnam (HMDC) from 1961 to 2017.

2.3. Data Analysis

As mentioned, this study applied the LVI and LVI-IPCC developed by [12] to calculate a composite LVI with weightings based on expert opinions and stakeholder discussions [17,34]. Calculation of LVI-IPCC is based on the IPCC definition [35], which defines livelihood vulnerability as a function of factors that contribute to exposure, sensitivity, and adaptive capacity. This then leads to proposals around adaptation. The methods for each of the vulnerability indices used in this study are provided in the following sections.

2.3.1. Composite Livelihood Vulnerability Index

We adapted the hierarchical approach [12] of constructing the LVI based on major components and associated subcomponents. In this study, the LVI has eight major components: (1) sociodemographic profile; (2) livelihood strategies; (3) social networks and finance; (4) health; (5) food; (6) water; (7) housing and productive land; and (8) natural disasters and climate variability. Compared to [12] and [30], we added a new major component—"Housing and productive land" in this research due to expected vulnerability, based on regional experience, related to injury/death as well as property damage/loss during extreme weather events. Furthermore, each major component is divided into specific sub-components (see Appendix A). Based on a review of existing literature, a field survey and consultations with numerous experts and local officials, 39 subcomponents (see Appendix A) were selected to assess the vulnerability level under the impact of climate change.

A balanced, weighted-average approach was employed to calculate the composite LVI [34]. Equal weighting of components was used in the absence of compelling evidence of a need and basis for differential weightings. Using equal weighting also makes the interpretation process simpler. This does, however, mean that while each subcomponent contributes equally to the overall vulnerability index, there is a difference in the number of subcomponents so that each major component contributes a different weighting to the overall vulnerability rating. Therefore, it is important to look closely at the subcomponent results. Data for the composite LVI are from household surveys, with the addition of regional precipitation and temperature data. The survey work was approved by the Human Research Ethics Committee of the University of Southern Queensland (approval for H18REA267).

Step 1: As many subcomponents are measured using different scales, e.g., numbers and percentages, each subcomponent needs to be standardized for comparability among them and for compiling the overall index as follows Equation (1).

$$Index_{SC_c} = \frac{SC_c - SC_{min}}{SC_{max} - SC_{min}}$$
(1)

where SC_c is the actual value of a subcomponent for a community c; SC_{min} and SC_{max} are the minimum and maximum values of each subcomponent reflecting low and high vulnerability, respectively.

Step 2: An index for each major component of vulnerability is then created by averaging their respective standardized subcomponents given by Equation (2).

$$MC_c = \frac{\sum_{i=1}^{n} Index_{SC_ci}}{n}$$
(2)

where MC_c represents each major components (eight major components) of the commune; and $Index_{SCc}^{i}$ is the indexed subcomponent value of each major component MC_c for the commune and n is the number of subcomponents in each major component.

Step 3: Once values for each of the major components for a community are calculated, they are averaged to obtain the community-level LVI, given by Equation (3).

$$LVI_{c} = \frac{\sum_{i=1}^{8} w_{MC_{i}}MC_{ci}}{\sum_{i=1}^{8} w_{MC_{i}}}$$
(3)

where LVI_c is the LVI for a community c which is the weighted average of the eight major components. The weights of each major component, w_{MCi} are determined by the number of subcomponents making up each major component and are included to ensure that all subcomponents contribute equally to the overall LVI (see Appendix B for an example of an LVI calculation). After calculating the major components and the LVI for each group, a spider diagram was also created to compare the vulnerability level in each major component among the groups. The LVI was scaled in the range from 0 to 0.7. A higher value for the LVI denotes more vulnerable systems. 2.3.2. Livelihood Vulnerability Index by IPCC Framework Approach (LVI-IPCC)

The LVI-IPCC is also calculated, with three major components: exposure, adaptive capacity, and sensitivity. This approach diverges from the composite LVI in how the eight major components of LVI are combined (Table 2). There are three steps to complete the LVI-IPCC computation (see Appendix C), including the inverse of subcomponents for adaptive capacity; the grouping of indicators; and the calculation of LVI-IPCC (detailed below).

Table 2. The contribution of th	: LVI eight major com	ponents to the LVI-IPCC
---------------------------------	-----------------------	-------------------------

LVI Major Components	IPCC Definition of Vulnerability (LVI-IPCC)	
Natural disaster and climate variability	Exposure	
Socio-demographic profile		
Livelihood strategies	Adaptive capacity	
Social network and finance		
Health		
Food	Samultivity	
Water	Sensitivity	
Housing and productive land		

Step 1: The same subcomponents as with the previously described approach were used. However, to fit with the IPCC's definition of vulnerability, the inverse of all subcomponents for adaptive capacity was calculated and then averaged in the relevant major components (see Appendix C for example).

Step 2: The IPCC-defined contributing factor of each category for a community $c(CF_c)$ was calculated by Equation (4)

$$CF_{c} = \frac{\sum_{i=1}^{n} w_{MC_{i}}MC_{ci}}{\sum_{i=1}^{n} w_{MC_{i}}}$$
(4)

where w_{MCl} is the weight of each major component and MC_{cl} is the index of a major component for a community c, and n is the number of major components in each contributing factor.

Step 3: After calculating the contributing factors, the LVI-IPCC index was derived using a linear function Equation (5)

$$LVI-IPCC_c = (CF_{c,exposure} - CF_{c,adaptive capacity}) * CF_{c,sensitivity}$$
 (5)

After calculating contributing factors and LVI-IPCC, these results were represented in vulnerability spider web diagrams for convenient visual comparison of the four groups. Each vertex of a web shows a contributing factor that can highlight differences between groups. The calculated values of the LVI-IPCC index represent the vulnerability level of each community, ranging from -1 to 1, i.e., from least to most vulnerable.

3. Results

3.1. Overview of Sample Groups

As expected, there were similarities and some differences between the ethnic groups as shown in Table 1. For similarities, all groups were highly dependent on their own farming for food, social network indicators were similar and more than 99% of all respondents could access information by television, radio, mobile phone, or internet. All groups were highly dependent on agriculture as the main source of income, with Muong having the
lowest level of dependency. Most households reported that they had not had access to any training related to climate change preparedness. The Hmong and Dao groups are, on average, further from the district centre, younger, have larger families, are more dependent on agriculture, and have more people with very low education levels.

3.2. Differences between Groups by LVI Components

Based on the eight major components and the composite LVI, there were differences between groups (see details in Table 3, with scoring differences in components shaded). The Hmong and Dao groups had higher overall LVIs and were more vulnerable on all component scores, with those differences in scores being a function of particular but not necessarily common (between these two groups) sub-components.

Components (Major Components in Bold)	Thai	Muong	Dao	Hmong
Dependency	0.342	0.319	0.378	0.396
Female headed household	0.247	0.175	0.182	0.107
Household heads did not attend school	0.548	0.546	0.591	0.679
Socio-Demographic Profile	0.379	0.347	0.384	0.394
Household mainly income dependent on agriculture/forestry (cultivation,	0.925	0.897	0.955	0.929
livestock, aquaculture, forest products collection)	0.007	0.000	0.545	0.571
Households without family members working outside the community	0.387	0.392	0.545	0.571
Households without non-agricultural livelihood income contribution	0.215	0.165	0.227	0.536
Average agricultural livelihood diversity index	0.165	0.160	0.190	0.263
Livelihood Strategies	0.423	0.403	0.479	0.575
Household without access to information (TV/radio/telephone/internet)	0.000	0.000	0.000	0.071
Average media diversity index	0.387	0.361	0.409	0.905
Need for assistance from the local government in last 12 months	0.387	0.402	0.545	0.679
Average receive/give ratio	0.156	0.122	0.080	0.138
Average borrow/lend ratio from/to the community	0.380	0.414	0.360	0.476
Average borrow/lend ratio from/to the bank	0.545	0.612	0.818	0.810
Average distance to the district centre	0.109	0.247	0.504	0.279
Households did not receive any agricultural training	0.570	0.505	0.682	0.786
Households did not receive any climate change training course	0.957	0.979	0.955	1.000
Households without any family member being a member of a group	0.140	0.113	0.227	0.429
Social Networks and Finance	0.363	0.376	0.458	0.557
Households with a family member with chronic illness	0.118	0.124	0.000	0.000
Households with a family member had to miss work or school in the last 2 week	0.022	0.093	0.182	0.143
une to inness Household with mombars modine daily domendant one	0.259	0.269	0.400	0.464
Avarana distance to account to builth captor (or bosnital)	0.256	0.205	0.409	0.404
Average distance to access to hearth center (or hospital)	0.100	0.165	0.007	0.317
Heath	0.142	0.162	0.299	0.281
Figure and the figure	0.925	0.959	0.955	0.964
Average number of months household struggle to find food for the family	0.000	0.062	0.114	0.070
Average Crop Diversity index	0.187	0.138	0.209	0.207
Household without crops saving	0.022	0.010	0.318	0.250
Food	0.300	0.292	0.399	0.373
Households utilize mainly natural water resources for domestic use	0.290	0.680	1.000	1.000
Average time to main water supply resource	0.016	0.067	0.030	0.029
Households do not have enough water for domestic use for the whole year	0.355	0.392	0.545	0.643
Inverse of the average days of stored water per household	0.528	0.562	0.497	0.693
Water	0.298	0.425	0.518	0.591
Households with weak thunderstorm/hail resistant construction	0.215	0.268	0.409	0.750
Houses elevated by low ground and easily inundated by floods	0.065	0.175	0.091	0.107
Houses is located at the place prone to a landslide	0.086	0.062	0.227	0.107
Average time to get to the agricultural land	0.185	0.159	0.425	0.358
Average areas of agricultural land vulnerability to floods	0.059	0.017	0.126	0.028
Average areas of agricultural land vulnerability to droughts	0.113	0.019	0.044	0.084
Housing and Productive Land	0.121	0.117	0.220	0.239

Tal	ы	e	3.	C	'n	72	ř.	
_		-	***	-	•		•••	

Components (Major Components in Bold)	Thai	Muong	Dao	Hmong
Average number of natural disaster in the past 5 years	0.298	0.347	0.308	0.347
Average types of natural disasters happened in the past 5 years	0.497	0.562	0.500	0.580
Household with losses physical assets and agricultural production in the past 5 years	0.570	0.526	0.864	0.750
Household did not receive a warning about the pending natural disasters	0.226	0.278	0.409	0.643
Natural Disasters and Climate Variability	0.426	0.439	0.475	0.499
Overall LVI	0.320	0.334	0.413	0.455
			and the second sec	

The shaded cells show scoring differences amongst the groups.

3.2.1. Socio-Demographic Profile

The Hmong and Dao groups had higher dependency levels, with Hmong having the highest percentage of household heads who have not attended school, but with the percentage of young female-headed households being the lowest. Overall, the Hmong group was more vulnerable than the other three ethnic groups on the socio-demographic profile component.

3.2.2. Livelihood Strategies

Hmong and Dao groups showed greater vulnerability on the livelihood strategies, which, in this study, included growing crops, raising animals, and collecting natural resources. Hmong and Dao households employed fewer livelihood strategies. Furthermore, Thai and Muong households reported having a higher percentage of family members working outside the community than did Hmong and Dao households. Similarly, Thai, Muong, and Dao ethnic groups have more non-agricultural income sources than the Hmong ethnic group.

3.2.3. Social Networks and Finance

More than half of Hmong and Dao households reported that they approached their local government for assistance in the past 12 months, compared to one-third of Thai and Muong households. Approximately 45-50% of Thai and Muong households had attended agricultural professional training, compared to 20-30% for Hmong and Dao households. These latter two households reported that the ratio of borrowing money and lending at the local bank was higher than that of Thai and Muong households. Hmong and Muong households had a higher ratio of the frequency of borrowing to the lending of money from and to family and friends. That is, people in these two groups tended to borrow more often, relative to lending occurrences. However, the ratio of in-kind assistance from family and friends and providing assistance in the past month was quite similar in three ethnic groups, with Dao people having the lowest rate. Thai, Muong, and Dao ethnic groups reported participating in a social organization more than Hmong ethnic groups. Such organizations included Farmer's Union, the Women's Union, the Young's Union, Farmer Interest Group, and Agricultural Cooperative. However, Hmong and Dao households reported that they mostly did not attend any training related to agriculture production. While there were variations across the sub-components, the net effect was that Hmong and Dao households were more vulnerable than Thai and Muong households on the social networks and finance components.

3.2.4. Health

Hmong and Dao reported travelling, on average, much further to the nearest health facility than did Thai and Muong households, the latter two reported higher rates of chronic illness amongst family members. Hmong and Dao households had higher rates of at least one family member missing work or school due to illness in the past 2 weeks. These two groups also had higher levels of dependent family members. Overall, the Hmong and Dao groups had higher health vulnerability scores.

3.2.5. Food

On average, Dao households had the longest average periods of food shortages. Muong households reported storing more crops for the next season than Thai, Hmong, and Dao households. Therefore, food vulnerability scores for Hmong and Dao ethnic groups were higher for the other groups.

3.2.6. Water

All Hmong and Dao households reported using a natural water source such as water from ravines, springs, or rivers to cook with and consume, which is presumed to increase vulnerability due to risks associated with water supply in the rainy season and water quality. Hmong and Dao households diverted water from ravines to plastic tanks through small water pipes. So while the average time to obtain water is relatively low, in the rainy season, these pipes are often buried by rocks and soil from the top of the hills or the mountains. Consequently, households in these areas can experience a short period of water deficits for daily needs. Meanwhile, nearly 70% of Thai and Muong households reported getting water from a personal well or clean water sources. Thai and Muong households have a greater water storage capacity, and more of the associated households have enough water for daily activities. Overall, the water vulnerability score for Hmong and Dao households was much higher than that for Thai and Muong households.

3.2.7. Housing and Productive Land

The majority of residential houses were built without technical guidance or professional instruction on reinforcement to mitigate the effects of natural disasters [36] but Hmong and Dao peoples have higher vulnerability scores in the housing and productive land component. Hmong especially have a higher average rate of using materials that have a low resistance to storms and hail. These materials include bamboo or unstable wooden planks and fibre cement sheeting. In addition, the physical location of a household is one important indicator considering the distribution of climate extreme events, especially in remote and hilly areas [2]. For example, households located along the river or stream networks are considered to be more vulnerable to flash-flood and bank erosion. Additionally, households situated at foothills' edges are likely more vulnerable than others when landslides happen. There is a higher rate of Muong households with housing in areas susceptible to flooding, while Dao households are more likely to be in places prone to landslide than other groups. Most households reported their areas of agricultural land were more vulnerable to floods than droughts, except for Thai households, which were more vulnerable to droughts than floods.

3.2.8. Natural Disaster and Climate Variability

There is no difference by ethnicity in regard to opinions on the average frequency and types of natural disasters (floods, drought, landslides, and so on) that occurred in the study area over the past 5 years. As reported, there were around four types of natural hazards and an average of around 20-24 hazard events. In the study sites, all information related to warnings and risks is transmitted through different channels including the announcement by digital means (for example, television, radio, village speakers, or inperson public meetings in the village). More Hmong and Dao people reported that they did not receive any warning about the pending climate extreme events such as frost, heavy rain, thunderstorm, flash flood, or landslides. Therefore, when the above natural disasters happened, more Hmong and Dao people reported that their house/property/agriculture production was damaged than did Thai and Muong people. The variables for the period of 1961-2017 used to develop the climate change rating, included mean standard deviation (MSD) of monthly average maximum daily temperature and minimum daily temperature, MSD of monthly average rainfall, average numbers of hot days, cold days, and heavy rain days. Overall, regarding natural disasters and climate variability, the Hmong and Dao were more vulnerable than Thai and Muong households.

3.3. Comparing LVI Outcomes

As a result of the above components, Hmong and Dao ethnic groups had a higher overall LVI score than did Thai and Muong ethnic groups. The results of the major component calculations are shown in a spider web diagram (Figure 3). This diagram is based on 0.1 unit increments, from 0 (less vulnerable) at the centre of the web to 0.7 (most vulnerable) at the web edge. The diagram illustrates that overall, Hmong and Dao ethnic groups are more vulnerable than Thai and Muong to livelihood strategies, social networks and finance, health, food, water, housing, and productive land.



Figure 3. Vulnerability spider diagram of the major components of the composite LVI for four ethnic group.

3.4. Comparing the Groups with the LVI-IPCC Index

The LVI-IPCC estimation indicated similar rankings among the four minority ethnic groups with the IPCC vulnerability value being highest for Hmong, followed by Dao, Thai, and then Muong (Table 4). Vulnerability triangles (Figure 4) present the values of contributing factors to the overall results of the groups, including exposure, adaptive capacity, and sensitivity. In terms of exposure, Hmong and Dao groups are more vulnerable than Thai and Muong groups. Hmong and Dao ethnic groups are more sensitive to climate variability than the other two groups, driven by differences in the health, food, water, housing and productive land components, as above. Thai and Muong had a higher adaptive capacity than Hmong and Dao groups, concerning demographics, livelihoods, and social networks and finance components. The overall LVI-IPCC index suggests Hmong may be particularly vulnerable.

IPCC Contributing Factors to Vulnerability	Thai	Muong	Dao	Hmong
Adaptive capacity	0.469	0.483	0.434	0.340
Sensitivity	0.204	0.234	0.344	0.356
Exposure	0.426	0.439	0.475	0.499
LVI-IPCC	-0.009	-0.010	0.014	0.057

Table 4. LVI-IPCC contributing factors for four ethnic groups in Phu Yen district.



Figure 4. Vulnerability triangle diagram of the contributing factors of the LVI-IPCC for four ethnic groups.

4. Discussion

This study takes a case study approach to explore the importance of accounting for intersectionality in programs aimed at addressing vulnerability to climate risk in disadvantaged agriculturally dependent smallholder communities in the poor upland region of Vietnam. In total, we interviewed 240 households from four ethnic minority groups in the Son La Province. By using the LVI and LVI-IPCC approach, which has been successfully applied in different contexts [4,15,20,37,38], the results confirm that there are differences in vulnerability between four ethnic minority groups within the Phu Yen district, Son La province, in the NMR Vietnam, even though there is considerable disadvantage across the region and all groups. Generally, ethnic people living in the mountainous region have lower levels of education and income and poorer housing quality. This region has fragmented topography, highly remote areas, and roads in poor conditions, making transportation difficult to nearby cities or the centre of the district for shopping and to access health care services when needed, especially in the rainy season [39]. Additionally, ethnic households in this area are poor and mainly rely on their self-farmed production for daily meals. Variation in the level of vulnerability in the four studied ethnic minority groups living at different elevations indicated that livelihood vulnerability in the district is not the same and varies according to spatial distribution. This result is consistent with the work of [30] who found that livelihood vulnerability was not homogenous within the communities they studied.

In terms of adaptive capacity, differences in the index are largely driven by differences in diversity of sources of income, debt, agricultural training, and organization membership. All groups are highly dependent on agriculture, but the Hmong and Dao have a lower rate of off-farm income. That means they are most vulnerable to seasons and events that adversely affect production. Hmong and Dao peoples also rely more heavily on borrowed money, which implies a higher degree of financial hardship of the households, which potentially reduces the adaptive capacity of households in the face of adverse climatic conditions and events [40]. In order to improve households' capacity and reduce the vulnerable level in climate-changing conditions, local governments could facilitate diversification through the development of off-farm employment opportunities, value-added industries such as handicrafts, job migration schemes and small business training [41], with some of those strategies supported by concessional loans programs [40,42]. Social network ratings were found to be potentially important factors in the vulnerability of households, especially in rural and poor areas [43]. Important social organizations that provided livelihood support included the Farmer's Union, the Women's Union, the Youth Union, the Farmer Interest Group, and Agricultural Cooperative. These can provide useful information on agricultural practices/activities such as new varieties, pest and disease status, price changes, crop calendar alteration, and managing climate extreme events. These groups could also strengthen social capital.

There may also be disparities in information flows. From our survey, Thai and Muong households reported they received more advice/training on farming activities from local authorities than did Hmong and Dao households. Possible factors in this difference are relative education levels, dependency rates with consequent constraining effects on households, rate of organization membership (lower for Hmong and Dao households), and remoteness. Remoteness may also contribute to fewer training opportunities (supply-side) and difficulty in getting to those opportunities. Therefore, local government could make a priority for education/awareness programs for remote and vulnerable ethnic groups, focusing more on Hmong and Dao groups to enhance their adaptive capacity. Furthermore, improved information regarding climate impacts and mitigation strategies provided by local government/governmental organizations could increase adaptive capacity.

In terms of sensitivity, access to health services and water sources are other areas of difference and apparent disadvantage. Access to health services has previously been proposed as affecting the health status of households [12,15,22]. From the present study, Hmong and Dao peoples travelled three times longer than Thai and Muong peoples did to health services. Remote roads are often in poor condition and the lack of compounds the problem for Hmong and Dao communities [39]. In addition to road improvement, as above, the government could also reduce health vulnerability through educational programs, make greater use of visiting health professionals, and further develop water quality strategies.

Water availability is more likely to be threatened under climate variability [44]. The reliance of ethnic households on natural water sources for drinking and agricultural purposes indicates high sensitivity to climate variability and change, especially during drought or in the dry season [45]. Based on the survey data, Hmong, Dao, and Muong households could be more vulnerable to water-borne diseases, such as cholera, diarrhoea, and measles-related to low water quality, whereas Thai peoples seem to have safer water sources due to the installation of boreholes. The vulnerability of households to water availability has been observed to be affected by conflict over scarce resources [46,47], with climate change potentially exacerbating these conflicts. However, in the present work, although some Muong households reported water conflicts, especially regarding using water for agricultural production, these conflicts were generally solved peacefully.

Regarding exposure, reference [2] found that the location of households is a key factor influencing how they prepared for natural hazards events. In this regard, Hmong and Dao groups are significantly affected by climate change compared to Muong and Thai peoples. The reason for this might be that Hmong and Dao people are not necessarily within the audible range of loudspeakers for announcements, are somewhat disconnected from general media, or are too far from the sites of public meetings that provide warnings and preparedness information. Government responses could therefore include improving communication technologies and reach, as well as outreach on preparedness training. Some households are also more vulnerable due to the flimsiness of housing materials, so housing reinforcement programs could enhance living conditions and resilience, which in turn, might also help alleviate poverty [4].

Our findings and recommendations are expected to support the Vietnamese national climate change adaptation plan for 2021–2030 with a long-term vision by the year 2050 (Decision No.1055/QD-TTg, dated 20 July 2020). We suggest targeted interventions (e.g., infrastructure development for market access, education and training programs for vulnerable ethnic groups) for enhancing resilience and adaptive capacity of communities, economic sectors, and ecosystems most vulnerable to climate change and variability.

5. Conclusions

Climate variability and climate events have been increasing in frequency and intensity in the NMR of Vietnam, which affects both livelihoods and production activities of various ethnic minority groups, and those who have lower levels of education and income and poor housing systems are likely to be especially affected. This study is the first assessment of the vulnerability of Thai, Muong, Dao, and Hmong ethnic groups in this region, using the livelihood vulnerability index framework (composite LVI and LVI-IPCC). The overall indices revealed differences based on ethnicity/location, with Hmong and Dao being the most vulnerable groups. This shows the potential of the two methods to identify what might be critical factors in more or less vulnerability and adaptive capacity. We conclude that an analysis of the sub-components of LVI are critical to formulating highly targeted responses, especially where program resources are very limited.

Our findings identified education levels, diversity of income sources, agricultural training, and organizational membership as the most important factors influencing the households' adaptive capacity. The diversity of income among all groups is relatively low, with high dependence on agriculture, so there is a high exposure to climatic effects. We observed that while all four ethnic groups had relatively low education levels and high dependence ratios, Hmong and Dao were especially vulnerable on these sub-components. These then are likely constraints on people's ability to receive and understand information and policies from the local government. Becoming a member of a social or professional organization or network provides more opportunities to get information on agricultural practices/activities/natural hazards and also to strengthen the connections among communities. Access to health services and water resources could also be important, with deficits in these areas further increasing vulnerability to climate variability, change, and events. Housing location and construction also contribute to household vulnerability, especially concerning extreme weather events.

In order to reduce the vulnerable level of ethnic groups to climate change, the study provides the following recommendations which may be of interest to researchers working in other remote rural areas in other regions:

- In national and local adaptation planning, priority should be given to support the poorer communities (in our studies case, the Hmong and Dao ethnic communities) that are more vulnerable and have a low capacity to cope with climate change.
- It is essential to enhance literacy, especially amongst disadvantaged groups (Hmong and Dao ethnic groups in the current work). This solution is important because this would increase the effectiveness of training and education programs, especially with understanding threats to livelihood, including climate change, and better enable the transfer of technology.
- Governments could strengthen extension, through targeted programs and appropriately designed visual aids and materials. These will help in the adaptation of farm systems and disaster preparedness.
- Local governments could facilitate income diversification strategies, supported by training and concessional loans.
- The government could upgrade road infrastructure to link remote communities to larger towns and centres and water systems and treatment.

Finally, we reiterate that the subjective selection and weighting sub-components for major components in the LVI models, and its influence on the vulnerability of households or communes, can be a limitation of LVI methods [4,12,45]. This research suggests that effective identification of the sub-components could improve the precision of assessment of the vulnerability of livelihoods to climate change at the local or regional level. To achieve that goal, researchers need to have a deep understanding of local situations including the natural resources, livelihoods assets, social-economic aspects, and climate conditions. The results of LVI-IPCC models in this research recommend that researchers should use caution in case the scores of LVI are negative or counterintuitive (the adaptive capacity results are greater than the exposure results). Increasing sensitivity might reduce the overall level of vulnerability. Therefore, in this case, in applying the LVI-IPCC model, caution should be taken in suggesting the adaptation options to climate change.

Author Contributions: Conceptualization, V.T.T., G.C., S.M., and D.-A.A.-V.; formal analysis, V.T.T., D.-A.A.-V., and G.C.; methodology, V.T.T., G.C., S.M., and D.-A.A.-V.; writing—original draft preparation, V.T.T.; writing—review and editing, G.C., S.M., V.T.T., and D.-A.A.-V. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by an USQ-VIED PhD Scholarship from the Vietnamese Government and the University of Southern Queensland (USQ). The APC and data collection expenditures were funded by the German Federal Ministry for the Environment, Nature Conservation, Building and Nuclear Safety (BMUB). All statements are the views of the authors and not the opinions of the funders or the Vietnamese Government.

Institutional Review Board Statement: The study was conducted according to the guidelines of the Declaration of Helsinki, and approved by Human Research Ethics Committee of the University of Southern Queensland (H18REA267, dated on 20 November 2018).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Data available on request due to restrictions.

Acknowledgments: The authors gratefully appreciate the contribution of all interviewed participants, and especially thank to the assistance of the local officers and colleagues in the Faculty of Environment Science, Vietnam National University of Agriculture for conducting the household survey. We also gratefully acknowledge the funding received from the German Federal Ministry for the Environment, Nature Conservation, Building and Nuclear Safety (BMUB).

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Table A1. Major components and subcomponents of the composite LVI and information sources.

Major Components	Subcomponents	Explanation of Sub-Component	Data Source	Explanatory Notes
	Female household head	Percentage of households where the primary adult is female. Women are usually more vulnerable than men.	Survey	Adapted from [15,22]
Socio- demographic profile	Dependency ratio	Percentage of household members who are outside employment age (under 15 and over 60 years old) as specified in Vietnam Labour Laws).	Survey	Modified from [12]. Modified the dependent age range to suit to the context of the study area
	Household heads did not attend school	Percentage of households where the head of the household reports that they have attended 0 years of school.	Survey	Adapted from [15,22]

Major Components	Subcomponents	Explanation of Sub-Component	Data Source	Explanatory Notes
	Households without family members working in a different community Households income	Percentage of households that report no family member working outside of the community for their primary work activity.	Survey	Adapted from [12,22
Livelihood	depends on agriculture/ forestry (cultivation, livestock, fishing, aquaculture, forest products collection)	Percentage of households that report only agriculture as a source of income.	Survey	Adapted from [12,33
strategies	Households without non-agricultural livelihood income contribution	Percentage of households that report no family member working in non-agricultural sector.	Survey	Adapted from [15]
Average livelihoo ir	Average agricultural livelihood diversity index	The inverse of (the number of agricultural livelihood activities +1) reported by a household, e.g., a household that farms, raise animals, and collects natural resources will have a Livelihood Diversification Index = $1/(4 + 1) = 0.2$.	Survey	Adapted from [12,33
Social networks and finance	Households without media access in the house	Percentage of households that report that they do not have any access to media information.	Survey	Adapted from [4]
	Need for assistance from the government in the last 12 months	Percentage of households that report that they have asked their local government for any assistance in the past 12 months.	Survey	Adapted from [12,22
	Average receive/give ratio	received by a household in the past month + 1) to (the number of types of help given by a household to someone else in the past month + 1).	Survey	Adapted from [12,22
	Average distance to the district's centre (e.g., km or minutes)		Survey	Adapted from [15,48
	Average borrow/lend money ratio (0.5-2)	satio of nouseholds borrowing money in the past month to a household lending money in the past month, e.g., If a household borrowed money but did not lend money, the ratio = 2:1 or 2 and if they lent money but did not borrow any, the ratio = 1:2 or 0.5.	Survey	Adapted from [12]
	Ratio of saving: saving at present (saving money in a bank + 1)/(borrowing money from a bank + 1)	Percentage of households that report that they do have not bank savings accounts.	Survey	New, added to reflec the context of study area: farmers often borrow money from a bank for agricultural production

Table A1. Cont.

		have all com.		
Major Components	Subcomponents	Explanation of Sub-Component	Data Source	Explanatory Notes
	Household without family members being a member of a cooperative society (such as women union, farmer union)	Percentage of head of households that report that any family member is a member of a cooperative society.	Survey	Adapted from [48]
	Households did not receive any climate change training	Percentage of the heads of household that report that they have not participated in climate change training.	Survey	Adapted from [34,49,50]
	Households did not receive training in their main profession/s	Percentage of the head of households that report that they have not participated in professional training	Survey	Adapted from [34,50
	Households with a family member having a chronic illness	Percentage of households that report at least 1 family member with chronic illness. Chronic illness was defined subjectively by the respondent.	Survey	Adapted from [12]
Health	Households with a family member missing work or school in the last 2 weeks due to illness	Percentage of households that report that at least 1 family member had to miss school or work due to illness in the last 2 weeks.	Survey	Adapted from [12]
	Average distance to health center (or hospital)	Average distance to the nearest health center (or hospital)	Survey	Adapted from [22,5]
	Households with members needing dependent care	Percentage of households that have at least one person needs to care for daily.	Survey	Adapted from [12,22
	Households primary dependent on self-farmed food	Percentage of households that get their food primarily from their personal farms.	Survey	Adapted from [15,33
Food	Average number of months households struggle to find food for the family (range: 0–12)	Average number of months households that struggle to obtain food for their family.	Survey	Adapted from [12,3
	Average Crop Diversity Index	fine inverse of (the number of crops grown by a household + 1). A household that grows pumpkin, maize, and cassava will have a Crop Diversity Index = 1/ (3 + 1) = 0.25.	Survey	Adapted from [12,3]
	Households that do not reserve crops	Percentage of households that do not save crops from each harvest.	Survey	Adapted from [12,30
	Households use mainly natural water systems for domestic use	Percentage of households that report a creek, river, lake, pool, or hole as their primary water source	Survey	Adapted from [12]
	Average time to main water supply resource	to travel to their primary water source.	Survey	Adapted from [12]
water	Not enough water for domestic use for the whole year	Percentage of households report that they do not have sufficient water to use for year-round activities	Survey	Adapted from [12]
	days of stored water per household	The inverse of (the number of days water stored + 1)	Survey	Adapted from [12]

Table A1. Cont.

Major Components	Subcomponents	Explanation of Sub-Component	Data Source	Explanatory Note
	Households with weak thunderstorm/ hail resistant construction	Percentage of households report that their house is susceptible to extreme weather events such as thunderstorm, hail, etc.	Survey	Adapted from [15
	Households on the low ground which is easily inundated by floods	Percentage of households that report that their house is easily inundated by flood	Survey	Adapted from [15
Housing and Productive Land	Households located at the places being prone to a landslide	Percentage of households with a house located in landslide-prone area.	Survey	New, added to emphasize the context of mountainous area which is more vulnerable to land slides
	Average time to get to the agricultural land	Average time it takes the household to travel to their agricultural land.	Survey	New, added to address the sensitivity of production land location to climate i
	Average areas of agricultural land being vulnerable to floods	Total area of household's agricultural land which is vulnerable to floods.	Survey	mountainous area Agricultural land located farther from the household is more difficult to tak care of the farm
	Average areas of agricultural land being vulnerable to droughts	Total area of household's agricultural land which is vulnerable to droughts.	Survey	especially in the cas of extreme weathe events
	Average number of natural disaster in the past 5 years Average types of	Total number of extreme weather events that were reported by households in the past 5 years.	Survey	Adapted from [12,4
	natural disasters happened in the past			Modified from [12,48].
Natural disasters and Climate Variability	Household with losses physical assets and agricultural production in the past 5 years	Percentage of households report that they had property loss and production because of extreme weather events in the past 5 years.	Survey	Modified from [15] Agricultural production losses would directly impact on livelihoo so it was added beside physical asse
,	Households with injury or death from natural disasters in the past 5 years	Percentage of households that report either an injury to or death of one of their family members as a result of the most severe flood, drought, or cyclone in the past 5 years.	Survey	Adapted from [1]
	Households did not receive a warning about the pending natural disasters	Percentage of households that did not receive a warning about the most severe flood, drought, and cyclone events in the past 5 years.	Survey	Adapted from [12]

Table A1. Cont.

Tabl	le A	1. 0	ont

Major Components	Subcomponents	Explanation of Sub-Component	Data Source	Explanatory Notes
	Mean standard deviation of monthly average minimum daily temperature (1961–2017)	Standard deviation of the average daily minimum temperature by month between 1961 and 2017 was averaged.	Data obtained from the Phu Yen meteorological station and HMDC	Adapted from [12]
	Mean standard deviation of monthly average maximum daily temperature (1961–2017)	Standard deviation of the average daily maximum temperature by month between 1961 and 2017 was averaged.		Adapted from [12]
	Mean standard deviation of monthly average rainfall (1961–2017)	Standard deviation of the average monthly precipitation between 1961 and 2017 was averaged.		Adapted from [12]
	Average numbers of hot days (1961–2017) (t \geq 35 °C) Average numbers of cold days (1961–2017) (t \leq 13 °C) Average numbers of	Number of hot days per year (t \geq 35 °C) between 1961 and 2017 was averaged. Number of cold days per year (t \leq 13 °C) between 1961 and 2017 was averaged.		New, supported by decision 03/2020/QD-TTG that prescribes forecasting, warning and communication
	Average number of days with heavy rain (1961–2017) (≥50mm/day)	Number of days with heavy rain per year (≥50mm/day) between 1961 and 2017 was averaged.		of natural disaster promulgated by the Prime Minister of Vietnam

Appendix B

Table A2. Calculating	the livelihood	l strategies major cor	nponent for the LVI	for Thai ethnic	group
-----------------------	----------------	------------------------	---------------------	-----------------	-------

Subcomponent for Livelihood Strategies Major Component	Subcomponent Values	Max Subcomponent Value	Min Subcomponent Value	Index Value	Livelihood Strategies Major Component Value
Percentage of household mainly income dependent on agriculture (LV1)	92.47	100	0	0.925	
Percentage of households without family members working outside the community (LV2)	38.71	100	0	0.387	0.423
Percentage of households without non-agricultural livelihood income contribution (LV3)	21.50	100	0	0.215	
Average agricultural livelihood diversity index (LV4)	0.33	1	0.2	0.165	

Step 1: Calculating the index value of subcomponents (repeat for all subcomponents)

$$Index_{LV_1Thai} = \frac{92.47 - 100}{100 - 0} = 0.93$$

Step 2: Calculating the value of livelihood strategies major component (repeat for all major components

$$Livelihood_{Thai} = \frac{\sum\limits_{i=1}^{n} Index_{SC_{c}i}}{n} = \frac{LV_{1Thai} + LV_{2Thai} + LV_{3Thai} + LV_{4Thai}}{4} = \frac{0.925 + 0.387 + 0.215 + 0.165}{4} = 0.423$$

Step 3: Calculating LVI for Thai ethnic group

.

$$LVI_{c} = \frac{\sum_{i=1}^{c} w_{MC_{i}}MC_{ci}}{\sum_{i=1}^{8} w_{MC_{i}}} = \frac{3*0.379 + 4*0.423 + 10*0.363 + 4*0.142 + 4*0.300 + 4*0.298 + 6*0.121 + 10*0.426}{(3+4+10+4+4+6+10)} = 0.320$$

Table A3. Calculating the contributing factors and LVI-IPCC for Thai ethnic group.

Appendix C

Contributing Factors	Major Components	Major Components Value	Number of Subcomponents for Major Components	Values of Contributing Factors	LVI-IPCC Value
Exposure	Natural disaster and climate variability	0.426	10	0.426	
Adapting	Socio-demographic profile	0.621	3		
Adaptive	Livelihood strategies	0.494	4	0.469	
Capacity	Social network and finance	0.413	10		-0.009
	Health	0.142	4		
C	Food	0.300	4		
Sensitivity Water	0.298	4	0.204		
	Housing and productive land	0.121	6		

Step 1: Calculate the index value for subcomponent and major component as presented in Appendix A. However, we need to take the inverse of all subcomponents for the contributing factor adaptive capacity (Socio-demographic profile; Livelihood strategies; Social Network and finance) before doing the calculation following Appendix A. An example for taking the inverse of subcomponents for Livelihood strategies major component is given below.

Table A4. The inverse of subcomponents for Livelihood strategies major component.

Subcomponent for Livelihood Strategies Major Component (LVI Calculation)	Subcomponent Values for Thai Ethnic Group	Subcomponent for Livelihood Strategies Major Component (LVI-IPCC Calculation)	Subcomponent Values for Thai Ethnic Group
Percentage of household mainly		Percentage of household with	
income dependent on agriculture (LV1)	92.47	income no dependent on agriculture (LV ₁)	7.53
Percentage of households without		Percentage of households with	
family members working outside the community (LV ₂)	38.71	family members working outside the community (LV ₂)	61.29
Percentage of households without non-agricultural livelihood	21.50	Percentage of households with non-agricultural livelihood	78.50
income contribution (LV3)		income contribution (LV ₃)	
Average agricultural livelihood diversity index (LV ₄)	0.330	Average agricultural livelihood diversity index (LV ₄)	0.504

Step 2: Calculate the adaptive capacity value for Thai ethnic group, repeat for contributing factors: Sensitivity and Exposure

$$CF_{Thei,adptivecapacity} = \frac{\sum_{i=1}^{n} w_{MC_i} MC_{ci}}{\sum_{i=1}^{n} w_{MC_i}} = \frac{3 * 0.621 + 4 * 0.494 + 10 * 0.413}{17} = 0.469$$

Step 3: Calculate LVI_IPCC value (repeat for all of 3 ethnic groups)

LVI-IPCC_{Thal} = $(CF_{c,exposure} - CF_{c,adaptive capacity}) * CF_{c,sensitivity} = (0.426 - 0.469) * 0.204 = -0.009$

References

5

- Dasgupta, S.; Laplante, B.; Meisner, C.; Wheeler, D.; Yan, J. The impact of sea level rise on developing countries: A comparative analysis. Clim. Chang. 2009, 93, 379–388. [CrossRef]
- Few, R.; Tran, P.G. Climatic hazards, health risk and response in Vietnam: Case studies on social dimensions of vulnerability. Glob. Environ. Chang. 2010, 20, 529–538. [CrossRef]
- Eckstein, D.; Künzel, V.; Schäfer, L. Global Climate Risk Index 2018; Germanwatch: Bonn, Germany, 2017. Available online: https://germanwatch.org/en/download/20432.pdf (accessed on 20 December 2018).
- Panthi, J.; Aryal, S.; Dahal, P.; Bhandari, P.; Krakauer, N.Y.; Pandey, V.P. Livelihood vulnerability approach to assessing climate change impacts on mixed agro-livestock smallholders around the Gandaki River Basin in Nepal. Reg. Environ. Chang. 2016, 16, 1121–1132. [CrossRef]
 - Chaudhry, P.; Ruysschaert, G. Climate change and human development in Vietnam. Hum. Dev. Rep. 2007, 2008, 2.
- Lee, S.; Dan, N.T. Probabilistic landslide susceptibility mapping in the Lai Chau province of Vietnam: Focus on the relationship between tectonic fractures and landslides. *Environ. Geol.* 2005, 48, 778–787. [CrossRef]
- UNDP. Viet Nam Special Report on Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation; UNDP: Hanoi, Vietnam, 2015. Available online: http://www.vn.undp.org/content/vietnam/en/home/library/environment_climate/ viet_nam_special_report_on_managing_the_risks_of_extreme_events_and_disasters.html (accessed on 25 June 2018).
- ISPONRE. Vietnam Assessment Report on Climate Change; Institute of Strategy and Policy on Natural Resources and Environment (ISPONRE): Hanoi, Vietnam, 2009.
- Cutter, S.L.; Mitchell, J.T.; Scott, M.S. Revealing the vulnerability of people and places: A case study of Georgetown County, South Carolina. Ann. Assoc. Am. Geogr. 2000, 90, 713–737. [CrossRef]
- 10. Prowse, M.; Scott, L. Assets and adaptation: An emerging debate. IDS Bull. 2008, 39, 42-52. [CrossRef]
- Dulal, H.B.; Brodnig, G.; Thakur, H.K.; Green-Onoriose, C. Do the poor have what they need to adapt to climate change? A case study of Nepal. Local Environ. 2010, 15, 621–635. [CrossRef]
- Hahn, M.B.; Riederer, A.M.; Foster, S.O. The Livelihood Vulnerability Index: A pragmatic approach to assessing risks from climate variability and change—A case study in Mozambique. *Glob. Environ. Chang.* 2009, 19, 74–88. [CrossRef]
- Pandey, R.; Bardsley, D.K. Social-ecological vulnerability to climate change in the Nepali Himalaya. Appl. Geogr. 2015, 64, 74–86. [CrossRef]
- Salik, K.M.; Jahangir, S.; ul ZafarZahdi, W.; ul Hasson, S. Climate change vulnerability and adaptation options for the coastal communities of Pakistan. Ocean Coast. Manag. 2015, 112, 61–73. [CrossRef]
- Shah, K.U.; Dulal, H.B.; Johnson, C.; Baptiste, A. Understanding livelihood vulnerability to climate change: Applying the livelihood vulnerability index in Trinidad and Tobago. Geoforum 2013, 47, 125–137. [CrossRef]
- Brooks, N.; Adger, W.N.; Kelly, P.M. The determinants of vulnerability and adaptive capacity at the national level and the implications for adaptation. *Glob. Environ. Chang.* 2005, 15, 151–163. [CrossRef]
- 17. Vincent, K. Uncertainty in adaptive capacity and the importance of scale. Glob. Environ. Chang. 2007, 17, 12-24. [CrossRef]
- Eakin, H.; Bojórquez-Tapia, L.A. Insights into the composition of household vulnerability from multi-criteria decision analysis. Glob. Environ. Chang. 2008, 18, 112–127. [CrossRef]
- Chambers, R.; Conway, G. Sustainable Rural Livelihoods: Practical Concepts for the 21st Century; Institute of Development Studies: London, UK, 1992.
- Nguyen, D.C.; Vo, H.T.; Chu, T.H. Application of Livelihood Vulnerability Index to Assess Risks from Flood Vulnerability and Climate Variability—A Case Study in the Mekong Delta of Vietnam. J. Environ. Sci. Eng. A 2013, 2, 476.
- Legese, S.A.; Olutayo, O.A.; Sulaiman, H.; Rao, P. Assessing Climate Change Impacts in the Lake Tana Sub-Basin, Ethiopia Using Livelihood Vulnerability Approach. J. Earth Sci. Clim. Chang. 2016, 7, 368.
- Adu, D.T.; Kuwornu, J.K.M.; Anim-Somuah, H.; Sasaki, N. Application of livelihood vulnerability index in assessing smallholder maize farming households' vulnerability to climate change in Brong-Ahafo region of Ghana. *Kasetsart J. Soc. Sci.* 2018, 39, 22–32. [CrossRef]
- Huong, N.T.L.; Yao, S.; Fahad, S. Assessing household livelihood vulnerability to climate change: The case of Northwest Vietnam. Hum. Ecol. Risk Assess. Int. J. 2018, 1–19. [CrossRef]
- Morse, S.; McNamara, N. Sustainable Livelihoods Approach: A Critique of Theory and Practice; Springer Science & Business Media: Amsterdam, The Netherlands, 2013.
- 25. GSO. Results of Vietnam Household Living Standard Survey 2014; General Statistical Office: Hanoi, Vietnam, 2015.
- Tuyen, T.Q. Income sources and inequality among ethnic minorities in the Northwest region, Vietnam. Environ. Dev. Sustain. 2016, 18, 1239–1254. [CrossRef]
- World Bank. The Social Dimensions of Adaptation to Climate Change in Vietnam. Discussion Paper Number 12 December 2010; Economics
 of Adaptation to Climate Change: Washington, DC, USA, 2010.

- Ha, T.T. The Impacts of the Doi Moi Process on Communities and Forest Land Management in Three Northeastern Provinces in the Northern Mountainous Region of Vietnam; The Australian National University: Canberra, Australia, 2007.
- IMHEN (Vietnam Institution of Meteorology, Hydrology and Climate Change); UNDP. Vietnam Special Report on Managing Risks of Extreme Events and Disasters to Advance Climate Change Adaptation; Vietnam Publishing House of Natural Resources, Environment and Cartography: Hanoi, Vietnam, 2015.
- Baffoe, G.; Matsuda, H. An Empirical Assessment of Households Livelihood Vulnerability: The Case of Rural Ghana. Soc. Indic. Res. 2018, 140, 1225–1257. [CrossRef]
- Le Ngoc, L.; Martinez-Valle, A.; Parker, L.; Bourgoin, C.; Nguyen Thi, T.; Guerten, N.; Grosjeans, G.; Pham Anh, H.; Vu Cong, L.; Vuong Thuc, T.; et al. Pragmatic Economic Valuation of Adaptation Risk and Responses Across scales. Case Study in Vietnam; Working Paper No. 185; CGIAR Research Program on Climate Change, Agriculture and Food Security (CCAFS): Copenhagen, Denmark, 2016. Available online: www.ccafs.cgiar.org (accessed on 23 May 2019).
- 32. Phu Yen Office of Statistics. Socio-Economic Report of Phu Yen District; 2018. Unpublished Report (in the File of Tran Thanh Van).
- Masud, M.M.; Rahman, M.S.; Al-Amin, A.Q.; Kari, F.; Leal Filho, W. Impact of climate change: An empirical investigation of Malaysian rice production. *Mitig. Adapt. Strateg. Glob. Chang.* 2014, 19, 431–444. [CrossRef]
- 34. Sullivan, C. Calculating a Water Poverty Index. World Dev. 2002, 30, 1195-1210. [CrossRef]
- IPCC. Climate Change 2001: Impacts, Adaptation and Vulnerability; Contribution of Working Group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change Cambridge University Press: Cambridge, UK, 2011.
- Tran, T.; Tran, P.; Tuan, T.H. Review of Housing Vulnerability. Implications for Climate Resilient Houses; Discussion Paper series Sheltering from A Gathering Storm No.1; Institute for Social and Environmental Transition-International: Boulder, CO, USA, 2012.
- Tewari, H.R.; Bhowmick, P.K. Livelihood vulnerability index analysis: An approach to study vulnerability in the context of Bihar. J. Disaster Risk Stud. 2014. [CrossRef]
- Tjoe, Y. Measuring the livelihood vulnerability index of a dry region in Indonesia: A case study of three subsistence communities in West Timor. World J. Sci. Technol. Sustain. Dev. 2016, 13, 250–274. [CrossRef]
- Do, T.; Nguyen, C.; Phung, T. Assessment of Natural Disasters in Vietnam's Northern Mountains; Paper No.54209; Munich Personal RePEC Archive (MPRA): University Library of Munich: Munich, Germany, 2013.
- Abdul-Razak, M.; Kruse, S. The adaptive capacity of smallholder farmers to climate change in the Northern Region of Ghana. Clim. Risk Manag. 2017, 17, 104–122. [CrossRef]
- Tongruksawattana, S.; Junge, V.; Waibel, H.; Diez, J.R.; Schmidt, E. Ex-Post Coping Strategies of Rural Households in Thailand and Vietnam. In Vulnerability to Poverty; Klasen, S., Waibel, H., Eds.; Palgrave Macmillan: London, UK, 2013; pp. 216–257.
- Soubry, B. How Can Governments Support Adaptation to Climate Change by Small-Scale Farmers? A Case Study from the Canadian Maritime Provinces? SSRN Electronic J. 2017. [CrossRef]
- Nawrotzki, R.J.; Riosmena, F.; Hunter, L.M.; Runfola, D.M. Amplification or suppression: Social networks and the climate change—migration association in rural Mexico. Glob. Environ. Chang. 2015, 35, 463–474. [CrossRef] [PubMed]
- Misra, A.K. Climate change and challenges of water and food security. Int. J. Sustain. Built Environ. 2014, 3, 153–165. [CrossRef]
 Etwire, P.; Al-Hassan, R.; Kuwornu, J.; Osei-Owusu, Y. Application of livelihood vulnerability index in assessing vulnerability to
- climate change and variability in Northern Ghana. J. Environ. Earth Sci. 2013, 3, 157–170.
- 46. Gunasekara, N.; Kazama, S. Water conflict vulnerability of regions. Risk Water Resour. Manag. 2011, 347, 267-273.
- Gunasekara, N.K.; Kazama, S.; Yamazaki, D.; Oki, T. Water Conflict Risk due to Water Resource Availability and Unequal Distribution. Water Resour. Manag. 2014, 28, 169–184. [CrossRef]
- Alam, G.M.M.; Alam, K.; Mushtaq, S.; Filho, W.L. How do climate change and associated hazards impact on the resilience of riparian rural communities in Bangladesh? Policy implications for livelihood development. *Environ. Sci. Policy* 2018, 84, 7–18. [CrossRef]
- Pandey, R.; Jha, S. Climate vulnerability index-measure of climate change vulnerability to communities: A case of rural Lower Himalaya, India. Mitig. Adapt. Strateg. Glob. Chang. 2012, 17, 487–506. [CrossRef]
- Alam, G.M.; Alam, K.; Mushtaq, S. Influence of institutional access and social capital on adaptation decision: Empirical evidence from hazard-prone rural households in Bangladesh. Ecol. Econ. 2016, 130, 243–251. [CrossRef]
- Barua, A.; Katyaini, S.; Mili, B.; Gooch, P. Climate change and poverty: Building resilience of rural mountain communities in South Sikkim, Eastern Himalaya, India. Reg. Environ. Chang. 2014, 14, 267–280. [CrossRef]

This chapter explored the livelihood vulnerability at community level. Four different typical ethnic communities in the Northwest Mountainous Regions of Vietnam were selected for assessing the level of vulnerability by using LVI approaches. The assessment revealed differences according to ethnicity and location, for example, Hmong and Dao groups are more vulnerable compared to Thai and Muong groups. Further analyses based on household surveys will be undertaken to investigate what sub-groups may be less resilient and how their resilience levels differ between both ethnic groups and gender (Chapter 4).

CHAPTER 4: PAPER 2 - NUANCED ASSESSMENT OF LIVELIHOOD RESILIENCE THROUGH THE INTERSECTIONAL LENS OF GENDER AND ETHNICITY: EVIDENCE FROM SMALL-SCALE FARMING COMMUNITIES IN THE UPLAND REGIONS OF VIETNAM

Originally publish as: VT Tran, D-A An-Vo, S Mushtaq, G Cockfield, 2022. Nuanced assessment of livelihood resilience through the intersectional lens of gender and ethnicity: Evidence from small-scale farming communities in the upland regions of Vietnam. Journal of Rural Studies. https://doi.org/10.1016/j.jrurstud.2022.03.011. This article cannot be displayed due to copyright restrictions. See the article link in the Related Outputs field on the item record for possible access.

Indicators/gender	Male	Female	Chi-Square test
Formal education	90%	78%	χ ² =21.79, <i>p</i> =0.019
Health problem	46%	77%	χ ² =16.52, <i>p</i> =0.034
Crop diversity	95%	85%	χ ² =184.88, p=0.000
Farm equipment ownership	46.1%	20.2%	χ ² =25.15, <i>p</i> =0.000

Supplementary Table 1. Correlation between indicators of household livelihood capitals and gender.

Indicators/Ethnicity	Thai	Muong	Dao	Hmong	Chi-square test
Salaried jobs	75%	82%	45%	42%	$\chi^2 = 25.06, p = 0.000$
Bank loans	42%	32%	67%	72%	χ ² =20.08, <i>p</i> =0.018
Rent farming equipment	12%	20%	64%	62%	χ ² =43.53, <i>p</i> =0.000
Illiteracy rate	4%	20%	54%	64%	χ ² =78.77, <i>p</i> =0.000
Agricultural trainings	61%	55%	35%	32%	χ ² =8.20, <i>p</i> =0.043
Group membership	92%	93%	77%	40%	χ ² =54.77, <i>p</i> =0.000
Family size (average, people)	4.2	4.5	5.1	7.1	χ ² =14.98, <i>p</i> =0.020
Farm size (average, ha)	0.82	1.02	1.52	1.36	χ ² =29.99, <i>p</i> =0.000
No of farm crops (average)	4.52	4.46	3.59	4.39	$\chi^2 = 29.97, p = 0.000$

This chapter provided nuanced views on self-assessed livelihood resilience at the household level among ethnic communities living in the upland region of Vietnam. By using the household livelihood resilience approach, we observed that ethnicity seems to play a more important role than gender in governing differences in household livelihood resilience in the study region. The chapter also indicated the effectiveness of using the intersectional lens of gender and ethnicity in resilience assessment and resilience building. Based on these finding in the chapter, we conducted further analyses to investigate whether perceptions of climate variability and responses to climate change vary among ethnic communities in the study region. In addition, we identified some main factors contributing to adaptation choices (Chapter 5).

CHAPTER 5: PAPER 3 - FARMERS' PERCEPTIONS AND ADAPTATION PRACTICES TO CLIMATE VARIABILITY RISKS AND THEIR ASSOCIATED DETERMINANTS: EVIDENCE FROM SMALL-SCALE FARMING COMMUNITIES IN THE UPLAND REGIONS OF VIETNAM

Article III: VT Tran, D-A An-Vo, S Mushtaq, G Cockfield: Farmers' perceptions and adaptation practices to climate variability risks and their associated determinants: Evidence from small-scale farming communities in the upland regions of Vietnam. (Had been submitted Weather and Climate Extremes, 2022).

Abstract

Climate change is threatening the livelihoods and food security of lowincome smallholder farmers in Vietnam, who may have little capacity or preparedness to cope with it. Adaptation at the farm level is therefore an important strategy for reducing the impacts of extreme climatic events. This study explored the perceptions of climatic attributes and their induced risks, adaptation practices, and associated determinants for small-scale farming communities in the Northwest Mountainous Regions. 240 representative farm households were interviewed in the study region. Our findings, from statistical analyses, indicated a majority of the farmers observed changes in temperature and rainfall pattern – an increase in temperature (which is consistent with the actual trend) and an increase in the rainfall (which is generally against the measured trend).

In addition, farmers reported climate-related risks – drought, flood, landslide, pests, and diseases, increasingly damaging crops. In response, farmers have adopted various strategies to respond to climate change such as changing crop varieties, crop diversification, crop management and protection (plant and soil), and finding off-farm jobs. However, there are statistically significant differences in adoption of each adaptive strategy to climate variability risks among studied small-scale farming communities. Furthermore, a binary logistic analysis showed that farmers' socio-economic factors, farm size, and institutional characteristics significantly influence households' adaptation. Future policies (free literacy classes, concession loans, upgrading infrastructure, natural disaster warnings, etc.), therefore, should consider these main factors to facilitate appropriate interventions and holistic climate change adaptation strategies and enable sustainable farming households' livelihoods for local farmers.

1. Introduction

Climate change can have significant repercussions for agricultural production, particularly in developing nations, where small-scale farming households are disproportionately affected and increasingly exposed to natural hazards (Wheeler & Braun 2013; Altieri & Nicholls 2017; Li et al. 2017). Climate change presents in increased climate variability, which is predicted to worsen in the next decades (Pachauri et al. 2014), threatening current and future development efforts. In fact, climate change has a particularly negative influence on agricultural production in developing nations and mountain regions, especially within Southeast Asia countries (IPCC 2014; Ali & Erenstein 2017; Huong et al. 2017; Khanal et al. 2018).

Vietnam is especially prone to natural disasters and is one of the top five nations affected by climate change (Smyle & Cooke 2011). This is evident through increasing average temperatures, rising sea levels, and drier seasons. Natural hazards including cyclonic storms, flood, and droughts, have occurred more frequently and severely in Vietnam. For example, the annual average temperature has risen by 0.5°C to 0.7°C and sea level has increased about 20cm over the last 50 years (Government of Vietnam 2011). According to the IPCC (2018), in mountainous regions, climate hazards include flash floods, storms, temperature extremes, drought, erratic rainfall, freezes and frost. The Northwest Mountainous Regions of Vietnam is highly vulnerable to such hazards because of comparatively fragile ecosystems, complicated terrain, and unstable geology (Do 2013), meaning this region is one of the most vulnerable areas to a changing climate (PanNature 2017).

Climate change's negative effects on agriculture are likely to be unpredictable, with the potential to be disastrous (Mushtag et al. 2020). For example, rice yields might drop anywhere from 6% to 42% by 2050, while other crop yields could drop anywhere from 3% to 47% (Smyle & Cooke 2011). Furthermore, climate change has negative impacts on soil quality, livestock, and farmer health (IPCC 2014). Given the trajectory of greenhouse gas emissions and consequent climatic effects, climate change adaptation is increasingly studied and promoted as a vital response for a vulnerable sector (Nelson et al. 2007; Barros et al. 2014; Thornton & Comberti 2017). Adaptation strategies such as diversifying crops, altering crop varieties and changes in irrigation systems and practices (Abid et al. 2016) are critical responses playing an important role in agricultural sustainability. However, smallholder farmers in mountainous regions such as those of Vietnam may have relatively low adaptive capacity (Pham et al. 2020; Tran et al. 2021) and limited capacity to learn and apply modern techniques to adapt to climate variability (Le Dang et al. 2014; Huong et al. 2017). Hence, strengthening farmers' adaptive capacity and their adaptation skills are critical for enhancing livelihood and food security (Nelson et al. 2009; Niles et al. 2015). Farmers' knowledge, views, and attitudes toward climate change are determinants of adaptation decisionmaking, both in approach and outcomes. In the process of climate adaptation, knowledge, perceptions, and attitudes are all interconnected and interacting (Deressa et al. 2011; Schad et al. 2012; Soubry et al. 2020). Therefore, it is critical to comprehend local perceptions of climate change and other elements that influence farmers' awareness of the issue. Indeed, it is considered the first step of the adaptation process (Di Falco & Veronesi 2013). Research shows that improving farmers' adaptive capacity can help to mitigate climate change's negative consequences, protect poor farmers' livelihoods, and enhance any potential benefits (Wheeler et al. 2013). Farmers' behavioural responses to natural hazards and opportunities are however heavily influenced by their perceptions (Pham et al. 2019).

While there are numerous studies on farmers' perceptions of climate change, on contributing factors and on adaptation strategies employed by affected farmers in many parts of the world (Pham et al. 2019), there has been very little research comparing farmers' perceptions and adaptation in different communities in the same mountain range to a changing climate (Schneiderbauer et al. 2021), making it difficult to examine divergent perspectives and support targeted interventions for different communities, especially if there are differences in histories and cultures on top of geographical differences. There is a lack of knowledge and understanding about how farmers prioritise and express their willingness to adapt to natural hazards. In addition, field-based research is considered critical because it may pave the way for a more accurate understanding of adaptation strategies to be applied at the local level. This work focuses on four farming communities with different predominant ethnicities, namely Thai, Muong, Dao, and Hmong, in the Northwest Mountainous Regions of Vietnam (NRMs), with all communities heavily relying on agriculture for their livelihoods. To start to address the aforementioned research gap, this paper aims to: (i) examine farmer's perceptions and differences between them, (ii) explore the current coping practices and adaptation employed to mitigate the negative effects of climate variabilities, and (iii) determine factors influencing farmers' adaptation choices.

2. Methods

2.1 Study site

Phu Yen district (see Fig. 1) in Son La province in the NMRs was chosen as the research site because it is one of the five most impoverished districts of Son La province. The district was a target area for the Vietnam Government's special poverty reduction program (Program 135) from 2016 to 2020 (Tran et al. 2021). This program is the socio-economic development program for remote and isolated areas, and especially for mountainous communes (Decision 135/QD-TTg dated 31st July 1998). The total area of the Phu Yen district is 1,227 km², with a population of approximately 116,000, the majority of which are involved in small-scale subsistence agriculture. Regional products include rice, fruits, and crops, with production systems being vulnerable to droughts and/or hot west wind in the dry season, flash floods and landslides in the rainy season, as well as cold spells, and frost in winter (Phu Yen Office of Statistics 2018). Ethnic minorities of the district distribute in three sub-regions based on elevation. The Thai and Muong groups reside at lower elevations, while Dao and Hmong groups settled, respectively, in the middle and high elevations (Dien 2002; Vien 2003).



Fig. 1. Mapping the study site (Phu Yen district) in Son La province. The pyramid provided the information on ethnic groups' location and their disadvantages.

2.2 Data collection

Primary data were collected through household surveys, using structured and semi-structured question frames, and key informant interviews (KIIs). A draft questionnaire was tested with fifteen farm households and then revised for formal household interviews. A survey of 240 households was then conducted from late 2018 to early 2019. The aim of the household survey was to collect information on stressors to livelihoods, perceptions of changes in climatic attributes and climate-related hazards, climate change implications, and farmer response plans. Survey questions were developed with specialists' and local agricultural expert consultations, a literature review and field trips. Farmers' perceptions of changes in climatic attributes and climate hazards during the past decade were categorised by four responses in regard to those changes: increase, decrease, stable, and don't know (Table 2). The respondents were also asked about their adaptation and coping strategies. Household interviews were conducted by the lead author and one local assistant. Elders, head of marginalised communities, and district/commune level officers (from the disaster risk management department and the natural resources and rural development department) were key informants.

Prior to interviews at households, a consent form was read aloud, signed immediately and collected by the lead author when the interview was completed. Generally, interviews were conducted only with the head of the family but if he/she was not available, spouses or the primary labourer were interviewed. Each interview took 1.0 to 1.5 hours on average and was conducted in the Vietnamese language. The local people, including the various minority groups, have mostly used this language in daily communication. Data were entered, checked, and analysed within Excel software 2016. Data on daily minimum and maximum temperatures and daily precipitation were collected from the Phu Yen meteorological station and the Hydro-Meteorological Data Centre of Vietnam (HMDC) from 1961 to 2017.

2.3 Data analysis

Data were analysed in SPSS 27.0 to analyse and describe households' awareness of climate variability, climatic related risks, chosen adaptation options and associated determinants. In addition, data from KIIs were combined with quantitative data to compare perceptions of key informants with those of households.

A one-way analysis of variance (ANOVA) test was used to test if there were statistically significant differences in selecting adaption strategies among the studied communities. The one-way ANOVA helps to identify significant differences in means between farming communities in selecting adaptation/coping strategies. If this ANOVA analysis returns a statistically significant *p*-value (less than 0.05), the alternative hypothesis is accepted, indicating statistical differences between studied farming communities. This test only revealed an overall difference among the studied groups, not the difference between two groups. Thus, we conducted one-way ANOVA with a Tukey post-hoc test to pinpoint intergroup differences.

A binary logistic regression model was developed to quantify the determinants affecting farmers' adaptation strategies. The dependent variable in the current work is whether a farmer has 'adapted' or 'not adapted' any adaptation practices to climate change. Based on discussions with staff from the natural resources and rural development department, a literature review, and field observations, five adaptation practices were identified: varieties; crop diversification; changing crop water management; crop management (crop and plant); and finding off-farm jobs. The selection of adaptation practices was categorized by a dummy variable (0,1) indicating whether a farmer employed a specific adaptation strategy (1) or not (0). Table 1 provides descriptions of explanatory variables and their expected signs in the current work. Because dependent variables (employing an adaptation practice or not) are binary, we used binary logistic models to analyse the factors influencing farmers' decisions to adapt to climate change because this model has been used in numerous prior research (Huong et al. 2017; Trinh et al. 2018). Another reason is that results from binary logistic model are more precise than those from similar binary models (e.g., Linear Probability Model that shows a limitation regarding error term distribution abnormality and heteroscedasticity) (Khan et al., 2020). In the binary logistic model, the dependent variable becomes the logarithm of the odds as presented in Eq. (1):

$$\log_{e}\left[\text{odds} = \frac{P}{1-P}\right] = \beta_{0} + \beta_{1}X_{1} + \beta_{2}X_{2} + \beta_{3}X_{3}...\beta_{k}X_{k}, \qquad (1)$$

Where *P* is the probability of adaptation; X_i , $i \in \{1, 2, ..., k\}$, are the explanatory variables and β_i are regression coefficients of the explanatory variables; and β_0 is the constant term.

Explanatory	Description	Value	Expected
variables			sign
Age	Age of farmer	Years	+
Gender	Gender of the farmer	1 Male; 0 Female	±
Education level	Level of education	1 illiterate; 2 primary; 3	+
	(degree)	secondary; 4 high school	
		and higher	
Ethnic location	Elevation at which	1 Low (Thai & Muong); 2	±
	ethnic groups reside	Middle (Dao); 3 High	
		(Hmong)	
Household	The condition of	1 Yes; 0 No	+
condition (poor	household		
household)			
Farm size	Number of hectares	Hectares	+
	of agricultural		
	farmland		
Access to credit	If the farmer has	1 Yes; 0 No	+
	access to credit from		
	any sources		
Access to	If the farmer has	1 Yes; 0 No	+
extension	access to extension		
	services		

Table 1. Description of explanatory variables hypothetically affectingfarmers' adaptation choices.

Explanatory	Description	Value	Expected
variables			sign
Access to	If the farmer has	1 Yes; 0 No	+
irrigation	access to credit		
Access to	If the farmer gets	1 Yes; 0 No	+
weather	information about		
information	weather, climate		
	from any sources		
Farmers' group	If the farmer is a	1 Yes; 0 No	+
membership	member of a farmer's		
	group		

Multicollinearity significantly influences the estimation of the logit model's parameters (Midi et al. 2010). Here, the binary logistic regression was conducted after we examined the association or correlation between explanatory variables based on the variance inflation factor (VIF) and contingency coefficient (CC) (Cama et al. 2016). The VIF was calculated for a continuous variable being related to other continuous variables, given by Eq. (2)

$$VIF(X_i) = \frac{1}{1 - R_i^2}$$
(2)

where, R_i^2 is squared multiple correlation coefficient between a continuous variable X_i and other continuous variables. Meanwhile, the CC was computed to explore any relationship between discrete (dummy) variables and categorical variables that could cause multicollinearity or association problems, given by Eq. (3)

$$CC = \sqrt{\frac{\chi^2}{n + \chi^2}}$$
(3)

where χ^2 denotes the Chi-square value and *n* the sample size.

A CC value around zero means no association and a CC value of more than 0.65 means two variables are associated. A VIF value less than 10 means no multicollinearity among continuous variables and a VIF value more than 10 means there is multicollinearity among continuous variables.

3. Results

3.1 Results of testing multicollinearity between explanatory variables

The calculated values of the VIF for three continuous variables demonstrates that none of the variables have a value of 10 or higher. The contingency coefficient was also calculated to see how closely the discrete variables were related. The contingency coefficient was calculated based on the value of chi-square for one categorial variable and six discrete variables (dummy). The calculated values of CCs for the discrete (dummy) and categorical variables were low (CCs < 0.65), indicating that there was no issue with multicollinearity between the variables. Therefore, the binary logistic model accepted all dummy and categorical variables.

3.2 Farmers' perceptions on climate variability and weather extreme events

Having a better understanding of farm households' perceptions of uncertain environmental conditions is crucial for suggesting changes to their farming activities. Farmers were asked to share their observations regarding climate variability and extreme events. Table 2 presents farmers' perspectives of changes in climate variables and extreme weather events through the four categories.

Table 2. Farmers' perception of climate attributes and extreme eventsthrough four categories: Increasing, Decreasing, Stable and I don't know

Climate	Climatic	Perceived	Respondents (%) agreement in			
Parameters	Indicators	trend	farming communities *			
			Thai	Muong	Dao	Hmong
			(n=93)	(n=97)	(n=22)	(n=28)
Precipitation	Overall rainfall	Increasing	84.9	85.2	83.5	81.3
		Decreasing	10.8	10.7	14.4	15.4
		Stable	4.3	5.1	2.1	3.3
		I don't know	0	0	0	0
Temperature		Increasing	83.9	81.8	78.4	64.3

	Local atmosphere	Decreasing	10.8	4.5	8.2	17.9
	temperature	Stable	5.4	13.6	13.4	17.9
		I don't know	0	0	0	0
Extreme	Frequency of	Increasing	81.7	81.8	83.5	82.7
events	flood and	Decreasing	2.2	9.1	7.2	7.1
	landslide	Stable	11.8	9.1	8.2	6.6
		I don't know	4.3	0	1.0	3.6
	Annual damaged	Increasing	15.1	22.7	32	39.3
	cold days	Decreasing	75.3	72.7	50.5	53.6
		Stable	9.7	4.5	17.5	7.1
		I don't know	0	0	0	0
	Annual extremely	Increasing	82.8	81.8	69.4	64.3
	hot days	Decreasing	9.7	4.5	8.2	7.1
		Stable	7.5	13.6	22.4	21.4
		I don't know	0	0	0	7.1
	Frequency of	Increasing	39.8	43.6	27.8	25.5
	drought	Decreasing	7.5	8.2	12.4	15.7
		Stable	49.5	48.2	59.8	59.3
		I don't know	4.2	0	0	0

*Due to financial resources available and time constraint for PhD study, small sample sizes used for the analysis of data for four studied ethnic groups. However, we used Slovin's formula to estimate the appropriate sample sizes for data collection. Please see Chapter 4 for more details.

Most of the respondents interviewed in various ethnic communities reported an increase in temperature. The outcome, however, differed among farming communities. Thai and Muong had a slightly higher proportion (above 81%) of farmers who thought the temperature had risen, compared to 78.4% and 64.3% of Dao and Hmong people (at medium and high altitudes), respectively. These perceptions of increasing temperatures were consistent with the actual regional trend (Fig. 2a). In contrast, there is high agreement among all studied farming communities about the trend of annual rainfall, e.g., above 81% of respondents perceived rainfall increases, which is somewhat against the observed trend of a slight decrease (Fig. 2b). Responses may have been affected by changes in the intensity of rainfall events (lower annual totals but higher falls per event). Respondents described how precipitation has been more intense and erratic. There may also have been a recency effect of heavy rain events, accompanied by flash floods that occurred in October 2017 and July 2018 in the study area, prior to the initial data collection. These flooding events greatly damaged local properties and crops and caused a few weeks of isolation and several deaths.

The frequency of extreme weather events is another important aspect of climate change. Flood, landslides, erratic rains, extreme temperatures, and droughts are the main extreme events occurring within the study region, which were reported by local farmers. A high percentage of farmers in all studied farming communities reported an increasing frequency of floods and landslides (from 81% to 84%, see Table 2). Similarly, an increase in annual extremely hot days (greater than 35°C) were reported by respondents. A higher proportion (above 81%) from Thai and Muong groups perceived that the numbers of annual extreme hot days had risen compared to 69.4% and 64.3% from Dao and Hmong groups, at the middle and high elevations. A decrease in annual damaging cold days were reported by 73% to 75% of Muong and Thai and 50% to 54% of Dao and Hmong people. The observed cold and hot day trends are in Fig. 2c and 2d. Views on drought frequency were more variable, though a majority of respondents opted for 'stable'. There were also some perceptions of seasonal shifts, as participants mentioned (e.g., Thai and Muong groups) that rainy seasons had typically lasted from June to September, but they had become more irregular recently. In addition, water reservoirs were reported to have dried up early because of insufficient rains and high evaporation.







3.3 Farmer's perception of climate and non-climate stressors

The studied farming communities have reported an emphasis on changes in frequency and severity of natural hazards. Throughout discussions with farmers in different communities, agricultural officials, and key informants, climatic and socio-economic pressures were reported to be major factors influencing agricultural activities. The socio-economic stressors include a lack of land, seeds, machinery and technical equipment, weather information, family labour, local government support, and money (detailed in Fig. 3). Flood and landslides, erratic rainfall, droughts, frost and freezes, pests, and diseases were named by respondents as the top climaterelated stressors affecting crop production.

Farmers from all studied groups indicated that floods and landslides (from 82.8% to 96%), and drought (from 54.5% to 67.9%) were the two most significant climate-related stressors affecting crop production. Participants stated that they were unable to prepare their land for the following season due to the occurrence of drought. In addition, an insufficient or limited irrigation system is noted as one of the factors exacerbating farmers' vulnerability to rainfall irregularity during crop seasons. Farmers mentioned changing rainfall involves shorter duration but higher intensities. Dao and Hmong respondents mentioned that it is hard to anticipate current climatic conditions for farming activities by applying historical, indigenous knowledge due to climate change. A higher proportion of Hmong groups, who reside in the highest location, reported frost and freezes are also impacting adversely on their crops and livestock. Furthermore, erratic rainfall pattern was also a climate stressor impacting agricultural activities within the Dao community. Thai and Muong people shared that the changes in rainfall and extreme weather events lead to an increase in pets and diseases adversely impacted on crop production, especially in crop yield.

a) Farmers' perception of climate-related stress (%)



b) Farmers' perception of non-climate stress (%)







Fig. 3. Climatic and non-climatic related stress reported by percentage of farmers in the studied farming communities.

All farmers participating in current study reported lack of crop seeds is one of the top non-climatic related stressors. However, there are differences in the list of top main non-climatic stressors influencing agricultural activities identified by Thai- Muong group and Dao - Hmong group. For example, Thai and Muong farmers reported their agricultural production was affected by lack of family labour (64% and 67%) and small farming land areas (60%). Meanwhile, Dao and Hmong considered lack of money (both cash and borrowing ability, 73% and 87%), lack of farming equipment (68% and 82%) and insufficient climate information (51% and 64%) that largely impacted on their crop production, especially in a changing climate. Differences in farmers' perception of non-climate stressors might relate to different living zones and their assets and capital (such as, financial, human, and physical capitals).

3.4 Household's adaptation practices to climate variability risks

After exploring farmer's perception on climate variability and climate associated stressors, farmers were also interviewed about adaptation practices they currently employed to mitigate adverse climate impacts. The data show a variety of adaptation strategies employed to protect and improve local livelihoods under the impacts of climate change. The adaptation strategies adopted by the respondents include changing crop varieties, diversifying crops, water management, crop management and protection (soil and plant), and finding off-farm jobs. As seen from Table 3,
higher percentages of Thai and Muong respondents reported practicing the above adaption measures compared to those of Dao and Hmong farmers. This might be explained from the higher financial capabilities in Thai and Muong people (Tran et al. 2022). Table 3 shows that approximately half of Muong and Thai respondents reported they grew drought-tolerant or shortduration crops whereas only about 10% to 15% of Dao and Mong's respondents used this adaptive strategy. As stated in the interviews, the cultivation of improved seed varieties would help farmers minimise their potential losses caused by climate variabilities, only a third of Dao and Mong farmers would consider this adaptation, perhaps because of the costs of the seed.

Table 3. Adaptation	practices t	to climate	variabilities	among	farming
communities					

	Resp	ondents (F-	p-		
Type of adaptation measurement		gro				
	Thai	Muong	Dao	Hmong	value	value
Changing crop varieties (drought,	48.4	46.4	15.5	10.7	1 608	0.003
flood tolerant/short duration)					4.050	0.005
Crop diversification	79.6	72.2	35.2	30.5	9.340	0.022
Water management	9.4	8.2	6.1	5.3	2.345	0.123
Crop management and protection	86.6	87.2	36.4	32.1	14.004	0.000
(Soil and plant)					14.964	0.000
Finding off-farm jobs	75.3	82.3	45.2	41.9	9.173	0.000

Diversifying crops is another adaptation method employed by 72% to 80% of Thai and Muong, and about 30% to 35% of Dao and Hmong farmers. According to a local officer, 'cultivating a single crop like rice or maize tends to increase farmers' vulnerability to a changing climate'. Diversifying crops in the same or different cultivated areas is likely to lower the risk of crop failure under a changing climate. In the past, there were two rice seasons in a year, and therefore, agricultural land was left fallow. Farmers then started planting maize or other suitable vegetables during their spare time after harvesting to reduce soil erosion caused by extreme rains or floods. In hilly areas (e.g., in Dao and Hmong communities), farmers employed intercropping techniques for their crop production such as cassava was planted with teak trees at the early life stages of teak trees. Households also indicated that this plantation combination is one of their adaptation measures to cope with the negative impacts of increasing temperature on crop health and to improve soil structure. In addition, this practice increases plant coverage and reduces surface runoff and soil erosion during rainy season. As a result, soil fertility can be maintained and productivity per cultivated unit is improved.

Crop management and protection (soil and plant) was adopted by around 87% of both Thai and Muong farmers, followed by 36% of Dao and 32% of Hmong respondents. Flash floods and landslides were seen as the main factors causing soil degradation and out-breaks of crop pests within the study region, thus, to protect soil and plants, farmers turned more to pesticides, fertilizers, and adopted mulching techniques on their land, crops residues being left on the ground to prevent soil erosion and increase moisture retention. On the other hand, farmers tend to adopt numerous soil protection means when they have cultivated land adjacent to rivers or streams. For instance, constructing or renovating their field embankments, creating stone walls, and planting bamboo trees were commonly used to reduce or prevent potential damage from natural disasters like flash floods and landslides.

Less than 10% of respondents from different farming communities used water management measures due to natural and economic difficulties (Table 3). This was the only strategy for which no statistically significant difference were found between groups. Farmers from all farming communities stated that due to rainfall irregularities, local reservoirs do not always fill during the rainy season. Furthermore, building water tanks at the household level and pumping water from rivers with a generator is seen as expensive. The fuel generator and related supplies to pump water from streams/rivers or wells cost more than 15 million VND (about US\$700). Some water reservoirs in Thai and Muong community areas had dried up, as respondents observed during the fieldwork. Due to limited irrigation access, efficient water usage strategies such as small-scale irrigated channels and other agricultural water collection systems are not installed in Dao and Hmong communities or not working effectively within Thai and Muong farming communities. Future interventions, therefore, need to encourage farmers to use any existing water resources by offering appropriate technologies which help farmers apply water conservation measures effectively to cope with climate change.

Finding off-farm jobs is another adaptation strategy adopted by 75% to 82% of Thai and Muong households, followed by 45% of Dao and 42% of Hmong households. Farmers reported they found off-farm employment through seasonal migrating to the nearby urban centres or cities, external aid programs or new local enterprises (Table 3). Besides, farmers living near forests reported that those forests provided continuous energy sources for the household (e.g., tree leaves and wood) and collecting forest products may create an additional income source for farming communities.

To evaluate adaptation strategies adopted among studied farming communities, the current work employed ANOVA. Results from ANOVA are in Table 3. These reveal significant differences between farming communities in almost all adaptation practices, except for water management. However, the differences between intergroups for each adaption strategy were not revealed by ANOVA. This was identified by adding a Tukey post hoc test while running ANOVA analysis. We also regrouped four studied communities into three groups: group 1 (Thai and Muong), group 2 (Dao) and group 3 (Hmong) based on their living zone elevation. The results of Turkey post hoc test indicate statistically significant differences between group 1 and group 2 and group 3 in adoption of changing crop varieties, crop diversification and finding off-farm jobs as their adaptive strategies (p < 0.005). However, there are only statistically significant differences between group 3 in choosing

crop management and protection (soil and plant) as an adaptive practice (p < 0.005).

3.5 Determinants of farmers' adaptation strategies

A logistic regression model was developed to determine the major influential factors influencing farmers' adaptation decisions. Table 4 provides the statistical results of the logistic regression of determinants influencing farmers' adaptation choices. We interpreted our findings through regression parameters such as coefficient (β), significance level (p-value), and odd ratio (Exp (β)). In the case of continuous variables (such as age, education level, and farm size, etc.), a negative coefficient indicates the likelihood of adopting a strategy decreases when continuous variables increase a unit. On the other hand, a positive coefficient indicates the likelihood of adopting a strategy increases when continuous variables increase a unit.

The regression results (Table 4) show that three explanatory variables including age, ethnic location (here referred to as elevation of their living zones) and access to extension services have a negative relationship with all four adaptation options to climate change. In contrast, four explanatory variables including gender, education level, access to irrigation, access to credit and access to climate information have a positive relationship with the four mentioned adaptation strategies. For the three remaining independent variables including household condition, farm size, and farmers' group membership have mixed effects on farmers' adaptation strategies to respond to climate variability. The details of the main determinants effecting each adaptation strategy are presented below.

3.5.1 Changing crop varieties

Four explanatory variables indicate a significant impact on adoption of changing crop varieties as a climate change adaptation practice. Specifically, farmer's education level (p < 0.05), household's access to irrigation (p < 0.05) and access to credit (p < 0.05) have positive significant

impacts, meanwhile farmer's age has a negative significant impact on farmers' choice of changing crop varieties (p < 0.1). On the other hand, the remaining explanatory variables have no significant effect on households' selection of crop varieties to cope with the changing climate.

The value of Exp (β) indicates the likelihood of changing crop varieties decreases by a factor of 0.590 and 0.560, respectively, if the age of the farmer increases by one year. The finding implies that the older farmers are the less likely to adopt climate tolerant crop varieties for their farming (e.g., short duration, drought tolerant crops). The older farmers are with more experiences have the potential to increase production if they adapt. However, age may limit effectiveness in the field, while younger generations are leaving for off-farm jobs such as builders or workers at garment and shoe factories in district centres and a nearby city. Additionally, the average age of Thai and Muong household heads is almost 50 years (Table 1, page 23), which is higher than the average age of minority ethnic household heads in Thua Thien Hue province (almost 41 years) (Sen et al 2020). This aging issue is also happening for Kinh farmers in other regions such as Thua Thien Hue province (Sen et al 2020). Therefore, the old age seems to be an important stressor, especially for Thai and Muong groups in the study region. In contrast, the Exp (β) of the education variable reveals that the likelihood of adopting tolerant varieties increases by a factor of 1.297 when household head education increases by one degree. Farmers who have higher educational attainment are more likely to adapt to climate variability by using resistant seed types such as short duration or climate tolerant crop varieties. They are typically motivated by later technologies and new seeds cultivation.

Apart from farmers' education and age, institutional services such as access to irrigation and credit, showed a positive significant impact on their changing of crop varieties. Farm households with access to irrigation were 1.443 times more likely to change crop varieties than those with little or restricted access to irrigation. Likewise, the result from the logistic model indicates that households having access to credit and loan availability were 1.913 times more likely to change crop varieties than those that did not use credit services. Hence, credit availability and utilisation would improve farmers' capacity in choosing climate tolerant crop varieties for their production adapting to climatic risks.

3.5.2 Crop diversification

Four explanatory variables have significant impacts on farmers' adoption of crop diversification as their adaptive strategy. Specifically, gender (p < 0.01) was found having a positive significant effect in adopting crop diversification. Male household heads are more likely to select crop diversification by 2.147 times (Exp (β) = 2.147) than female-headed households. In contrast, the age of farmers (p < 0.01), ethnic location (p< 0.05), and access to extension services (p < 0.05) have negative significant effects on the crop diversification adoption. Farmers living in lowland areas (Thai and Muong groups) are less likely to choose this adaptation practice by a factor of 0.521 than farmers in middle and high land zones (Dao and Hmong groups). Likewise, farmers accessing extension services are less likely to choose crop diversification as an adaptation practice to climate variability. This negative relationship might reflect the local situation that farmers continue making their own farming decisions despite attending agricultural training or meetings providing information on crop diversification. Farmers suggest that the quality of current extension services is insufficient to effectively aid farmers in adaptation process because training documents are out of date (Tran et al. 2022). Some also think a high percentage of extension staffs at a young age, have insufficient experience in agricultural activities. Farmers would thus be reluctant to follow their advice to diversify their crops or other farming-related activities.

3.5.3 Crop management and protection (soil and plant)

Crop management and protection (soil and plant) is a popular strategy to cope with the currently changing climate. We found that access to climate information, farmers' group membership, and farm size have positive significant effects (p<0.05, 0.1 and 0.1, respectively) while access to extension services has a negative impact in farmers' adoption of crop management and protection.

Farmers having larger land sizes are more likely to select this adaptation practice (Exp(β)=1.788). This accords with the proposition that adaptation has a fixed cost, with the effect on average cost diminishing by increasing scale. Additionally, farmers with larger areas can more easily generate any capital needed to make changes (e.g., innovate irrigation systems). In the current study area, farmers applied more crop management and protection strategies (soil and plants) to cope with natural hazards (e.g., landslides, flood, and flash floods). Plant protection means pesticides and fertilisers. Soil conservation includes mulching techniques or reinforcing embankments, creating stone walls, and growing bamboo trees along/around farmland areas where easily vulnerable to floods and landslides. Farmers who received weather-related information through many sources (television, radio, internet, newspaper, village speaker, neighbours) were more likely by a factor of 1.754 ($Exp(\beta)=1.754$) to apply crop protection and soil management as responses to natural hazards. These responses are such as using pesticides and fertilizers, reinforcing embankments, creating stone walls, growing bamboo trees along/around farmland areas, and clearing irrigation channels before or after the occurrence of natural hazards. Similarly, being a member of farmer groups increases the probability of choosing crop management and protection (plant and soil) by a factor of 1.634.

3.5.4 Finding off-farm jobs

Finding off-farm jobs is a favourable adaptation strategy chosen by the local farmers to climate variability risks. This choice was significantly driven by ethnic location, farm size, and access to irrigation (p < 0.05). Access to irrigation has a positive effect while ethnic location and farm size show negative relationships in adoption of off-farm jobs. Farmers living in middle

and high land zones (Dao and Hmong groups) are less likely to choose finding off-farm jobs by a factor of 0.584 than who reside in lowland areas (Thai and Muong groups). Also, households with more land decrease the probability of selecting finding off-farm jobs as an adaptation strategy to the changing climate by a factor of 0.619 per ha. On the other hand, farmers who have access to irrigation systems are 1.90 times more likely to find offfarm jobs as a response to a changing climate than farmers who do not have access to irrigation. This might be explained by the fact farmers need more income to pay for their high irrigation costs. In particular, the very water resources on which irrigation depends might be dependent on rainfall, therefore, supply constraints and rising demand might also increase irrigation costs during dry seasons.

Table 4. The binary logistic regression model predicting determinantsof farmers' adaptation practices.

Variables	Changing crop		Crop diversification		Crop management		Finding off-farm jobs	
	varieties				and protection (soil			
					and plant)			
	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)
Age	-0.526	.590*	-0.580	0.560***	-0.210	0.810	-0.520	.594
	(0.271)		(0.22)		(0.150)		(0.21)	
Gender	0.270	1.310	0.764	2.147***	.0680	1.974	0.820	2.270
	(0.185)		(0.434)		(0.352)		(584)	
Education	0.260	1.297**	0.336	1.399	0.248	1.281	0.294	1.342
level	(0.102)		(0.203)		(0.254)		(3.17)	
Ethnic	-0.772	0.462	- 0.652	0.521**	-0.848	.428	-0.538	0.584**
location	(0. 363)		(0.539)		(0.312)		(0.312)	
Household	0.146	1.157	-0.882	0.414	0.608	1.837	0.367	1.443
condition	(0.188)		(0.584)		(0.368)		(0.227)	
Farm size	0.531	1.700	-0.329	0.720	0.581	1.788*	-0.480	0.619**
	(0.450)		(0.246)		(0.235)		(0.138)	
Access to	0.367	1.443**	0.601	1.824	0.621	1.861	0.642	1.900**
irrigation	(0.155)		(0.452)		(0.358)		(0.343)	
Access to	-0.581	0.560	-0.452	0.636**	-0.458	.632*	-0.572	0.564
extension	(0.235)		(0.222)		(0.250)		(0.349)	
Access to	0.649	1.913**	0.169	1.184	0.132	1.141	0.418	1.519
credit	(0.351)		(0.083)		(0.061)		(0.288)	
Access to	0.362	1.436	0.443	1.557	0.562	1.754*	0.321	1.378
climate	(0.225)		(0.204)		(0.242)	*	(0.156)	
information								

Variables	Changing crop		Crop diversification		Crop management		Finding off-farm jobs	
	varieties				and protection (soil			
					and plant)			
	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)	Coeff. (β)	Exp (β)
Farmers'	-0.233	0.792	-0.161	0.851	0.491	1.634*	-0.320	.726
group	(0.197)		(0.227)		(0.241)		(0.335)	
membership								
Model	-1.612	0.199	1.297	3.658	1.369	3.931	-1.149	0.317
Constant	(1.115)		(.844)		(0.932)		(1.375)	
Model	99.80***		132.38***		90.42***		91.58***	
chi-square								
-2Log	225.600		145.075		257.960		161.678	
likelihood								
Pseudo R ²	0.483		0.673		0.450		0.580	

*Note: The values in the brackets are Stand Errors; *, **, *** indicates significance level at 10%, 5%, 1%, respectively.*

4. Discussion

Comprehensive understanding of climate change and variability is crucial to determine preparedness and implementation of adaptation measures. It helps policymakers design policies and/or intervention assisting farmers for better adaptive strategies to mitigate the impacts of potential risks such as poverty and food insecurity. In line with previous studies in the Northwest Mountainous Regions of Vietnam, local farmers mostly perceived changes in temperature, rainfall, and extreme weather events (CARE 2013; UNDP, 2015; Huong et al. 2017; Nguyen et al. 2019). Increases in annual average temperature, flood and landslides, hot days and decreases in cold days are being noticed by individuals, in association with personal experiences. Increases in rainfall was reported by respondents, which may have been a recent effect of intensive rain events. This supports the idea that recent and direct experiences with extreme weather events have shaped locals' perception of climate change and variability, especially for memories of recent variability instead of long-term changes in climate (Le Dang et al. 2014).

Climate and non-climate stresses have negative impacts on livelihood sources within the study region, weakening adaptive capacity and household livelihood resilience to the changing climate (Nguyen & Leisz 2021; Pham et al. 2021). For example, labour shortages and limited farm inputs were identified as socioeconomic pressures in the studied communities for changing agricultural practices (Antwi-Agyei et al. 2018). In the present study, Thai and Muong participants shared that youth migration to cities or youth involvement in local industrial companies recently operating around their villages are main reasons for labour shortages in farming activities.

Changing crop varieties, crop diversification, water management, crop management and protection (soil and plant), and finding off-farm jobs are major adaptation strategies currently employed in the study area. The list of four mentioned adaptation strategies was also employed to adapt to climate variabilities at different locations but with a difference in the degree of adaptation practices (Bahinipati 2014; Masud et al. 2017; Fahad & Wang 2018; Harvey et al. 2018; Trinh et al. 2018; Abid et al. 2020).

The results of the binary logistic model indicate that age, ethnic location, farm size, access to irrigation, access to extension are important factors influencing farmers' decisions in adaptation to climate change and variability. Older farmers are less likely to employ adaptation measures such as changing crop varieties and diversifying crops to climate variability impacts. Abid et al. (2015), Pham et al. (2019) and Ojo and Baiyegunhi (2020) indicated that the relationship between farmers' age and their willingness to change/diversify crop varieties is negatively significant. Senior farmers are typically illiterate and cultivate based on their past farming habits. They are also risk averse (Khan et al. 2020) and thus less likely to try new crop varieties because of the risk of not achieving expected yields. Male respondents are more willing to adopt adaptive practices mitigating potential risks of climate change and variability than females, especially for crop diversification. Male headed household mostly reported greater crop diversities on their farms than female headed households (Trinh et al. 2018; Getahun et al. 2021; Bui and Do 2021; Tran et al. 2022). In the study area, men believed that they would achieve stable production and higher household incomes by diversifying their crops. On the other hand, we found that farmers having better education are more likely to employ adaptation strategies. Ashraf et al. (2014), Bahinipati (201) and Ngo (2016) indicate that education increased the likelihood of farmers adapting to climate variabilities by growing climate tolerant seeds and using irrigation systems. However, the percentage of local farmers with limited formal education ranges from 55% to 68% (Tran et al. 2021), therefore, there may be adaptation benefits in enhancing basic education levels, especially for women and disadvantaged groups. This can be done by organising free literacy classes or tutorial classes in consultation with farmers' customs.

Ethnic location is another factor affecting local farmers' decisions to cope with climate change and variability. Khanal et al. (2018) conducted a study in Bangladesh, indicating that farmers living on flat and low land are more likely to adapt to climate change impacts compared to farmers residing in hill and mountain regions. In the current work, farmers reside at low land areas (Thai and Muong groups) are more likely to employ adaptation strategies such as finding off-farm jobs and crop management and protection compared to farmers living in middle and high land (Dao and Hmong groups). Thai and Muong groups live near the district centre being more advantages to access to public facilities (health and education services) and labour markets compared to Dao and Hmong groups (Tran et al., 2021). Moreover, road systems within Dao and Mong communities have greater levels of deterioration compared to those for Thai and Muong communities (Tran et al. 2022). Local government thus needs to invest more on upgrading the road systems to facilitate communications and access to public services of remote communities (e.g., Dao and Hmong groups).

Farmers with large farmland are more likely to have relatively higher capital and other resources for climate change adaptation, employing more crop management and protection means. Piya et al. (2013), Ashraf et al. (2014), Jin et al. (2016), Ndamani & Watanabe (2016) and Ngo (2016) found that farmers having large farmland tend to employ adaptation strategies related to their farms. This might come from the fact that large farms are likely to suffer high economic losses caused by climate change and variability compared to small farms. In contrast, farmers with small land are more likely to find off-farm jobs to increase their incomes as a response to climate change impacts. Huong et al. (2017) found that farmers with larger farm sizes had lower probability to choose off-farm jobs as an adaptation measure and therefore have more labour hours to make and manage system changes.

Access to irrigation services is another influential factor in adaptation choices such as changing crop varieties and finding off-farm jobs. Our results are consistent with those found by Khanal et al. (2018) that irrigation accessibility is one of the main factors influencing farmers' decision to use enhanced crop varieties. Accessing irrigation service is more likely to encourage farmers to find off-farm jobs, which is similar to the finding of Pham et al. (2019). In the study region, although farmers had access to constructed irrigation systems, water from irrigation systems is insufficient for their farms, particularly during the dry season (winter season) (Tran et al. 2022). Therefore, upgrading/installing irrigation system in local communities should be a priority of local interventions to increase farmers' physical capital in response to climate variability.

Access to extension services, on the other hand, was identified to have a negative impact on farmers' adaptation choices such as crop diversification, crop management and protection (soil and plant). Bui & Do (2021) and Sertse et al. (2021), however, found a positive correlation between extension services with choosing adaptation strategies. In the current study, limited agricultural staff visits, the low quality of extension staff and training documents are the main reason for ineffective extension services. Therefore, there is a need to improve the quality of agricultural officials and training materials as well as to increase communication between farmers and extension workers at their houses or at monthly meetings.

Access to credit is another important factor influencing farmers' choices to adapt to climate change impacts. The coefficient of access to credit is positive for all adaptation strategies and significant only for changing crop varieties as adaptation strategies. This finding aligns with the past studies on a positive association between access to credit and changing crop varieties (Ojo & Baiyegunhi, 2020; Sertse et al., 2021). The availability of credit services and loans will support farmers in accessing new technology and purchasing improved varieties of seeds, fertilizers, and other requisite inputs to adopt any adaptation strategies. Additionally, with greater financial support, farmers can adjust their management practices to adapt to a changing climate. Local government could support farmers' adaptation by providing concessional loans and regular meetings to guide locals how to use these credit services.

Access to climate information has positive influence on choosing adaptive strategies, especially being significant with the choice of crop management and protection (soil and plant). Van et al. (2015) and Alam et al. (2016) also found positive relationships between access to climate information and adaptation practices. Farmers who are receiving weather information and other risks warnings are more likely to adopt adaptation strategies compared to those without climate information. However, in the study region, the percentage of Dao and Hmong who did not receive ontime natural disasters warning are double the percentage of Thai and Muong due to remoteness and financial conditions (Tran et al. 2021). Hence, local government should provide a speaker louder for each village or equip a computer with internet for head of village, which can enable transmitting climate weather report and natural disaster warnings to local farmers ontime.

Famers who participate in cooperative organisations have a higher probability to employ adaptive strategies to minimize the effect of climate change. The relationship between farmer group membership and the choice of crop management and protection (soil and plant) is positive and statistically significant. The positive correlation could be traceable from the fact that membership of farmer groups can be beneficial by sharing experiences, exchanging information about any innovations, and implementing collaborative agreements (Kassie et al. 2013; Ndamani & Watanabe 2016). Therefore, membership of farmer groups improves cognitive processes through interactions among members (Bandiera & Rasul 2006). However, the relationship between farmer group membership and other three adaption practices are negative and non-significant. Hence, there is a need to investigate factors contributing to ineffectiveness of farmer groups and how to improve the quality of these groups.

5. Conclusion and implications

This study aims to investigate ethnic farm households' perceptions of climate change, climatic stressors, adaptation choices, and their determinants. The study was conducted to assist the Vietnamese governments in developing more appropriate supporting policies for farming communities by providing an improved understanding of ethnic farm households' behaviours and responses when they decided to select adaptation strategies to climate change.

We found that the local climatic conditions are changing and already have had a significant negative impact on local people's livelihood resources. Farmers from all ethnic communities reported changes in climatic conditions and natural hazards, most notably, that precipitation has changed in terms of frequency and intensity, and annual temperature has increased. Farmers also reported climatic stressors have impacted agricultural activities. Farmers, as a response to climatic risks, adopted a variety of adaptation strategies to mitigate the negative effects of climate related hazards on their agricultural activities and livelihoods. The most common adaptation strategies include changing crop varieties, diversifying crops, soil and plant management and protection, and finding off-farm jobs. These adaptation practices were mostly practiced by Thai, Muong, and Dao farmers. There were statistically significant differences among studied farming communities when they select the above common adaptation practices. However, only water management practice does not statistically differ among ethnic farmers.

Climatic and non-climatic stressors have seriously impacted the livelihoods of local farming communities. Hence, it is critical to develop a comprehensive adaptation strategy to climate change, which considers both mentioned stressors in order to reduce vulnerability and increase farmers' adaptive capacity.

The regression analysis revealed that farm households' characteristics such as age, gender, ethnicity, and education level, have significant impacts on farmers' adaptation measure preferences. Our present study suggests more appropriate policies focusing on enhancing local farmers' knowledge and skills and investing in education systems. For example, free literacy classes and training programs related to upgraded technology promoting sustainable land use and cultivation. Additionally, providing agricultural production inputs (such as climate tolerant varieties) at reasonable prices for farmers or free seeds for the most disadvantaged ethnic group (i.e., Dao and Hmong) could improve their income, which enables better financial capital to cope with climate change impact. An-Vo et al. (2021) demonstrated that the use of those tolerant varieties with seasonal climate forecasts can return significant economic values to farmers. To encourage more farmers to plant the tolerant varieties, especially for the disadvantaged group, local authorities should reduce the price of seed varieties or provide first free seedlings, which in turn enable better farmers' crops under changing climate.

Furthermore, the current work found farm size and other institutional factors (i.e., irrigation systems and extension service connections, credit services, climate information, and farmer groups) could have a significant impact on farmers' adaptation choices. Future policy options include

enhancing the quality of extension workers/training documents, building/upgrading new/existing irrigation systems, and providing media equipment other associated technology regarding receiving and transmitting climate information. Local authorities should organise free training courses (e.g., cultivation and breeding techniques) to help commune extension staff improve their knowledge and skills or send commune staff for short professional courses held at collaborative universities. Additionally, extension officers should communicate with local farmers more frequently to build up a better relationship and gradually gain their trust. Local government also should provide farmers with information on weather and climate induced risks on time, which facilitate preparedness and prevention effectively.

References

- Abid, M., Scheffran, J., Schneider, U.A. & Ashfaq, M., 2015. Farmers' perceptions of and adaptation strategies to climate change and their determinants: the case of Punjab province, Pakistan, Earth Syst. Dynam., vol. 6, no. 1, pp. 225-43.
- Abid, M., Schilling, J., Scheffran, J. & Zulfiqar, F., 2016. Climate change vulnerability, adaptation and risk perceptions at farm level in Punjab, Pakistan. Sci. The Total Environ., vol. 547, pp. 447-60.
- Abid, M., Ali, A., Rahut, DB., Raza, M. & Mehdi, M., 2020. Ex-ante and ex-post coping strategies for climatic shocks and adaptation determinants in rural Malawi. Clim. Risk Manag., vol. 27, p. 100200.
- Alam, GM., Alam, K. & Mushtaq, S., 2016. Influence of institutional access and social capital on adaptation decision: Empirical evidence from hazard-prone rural households in Bangladesh. Ecol. Eco., vol. 130, pp. 243-51.
- Ali, A. & Erenstein, O., 2017. Assessing farmer use of climate change adaptation practices and impacts on food security and poverty in Pakistan. Clim. Risk Manag., vol. 16, pp. 183-94.

- Altieri, M.A. & Nicholls, C.I., 2017. The adaptation and mitigation potential of traditional agriculture in a changing climate. Clim. Chang., vol. 140, no. 1, pp. 33-45.
- An-Vo, D.-A., Radanielson, A., Mushtaq, S., Reardon-Smith, K. & Hewitt, C., 2021.
 A framework for assessing the value of seasonal climate forecasting in key agricultural decisions. Clim. Serv. 22, 100234.
- Antwi-Agyei, P., Dougill, A.J., Stringer, L.C. & Codjoe, S.N.A., 2018. Adaptation opportunities and maladaptive outcomes in climate vulnerability hotspots of northern Ghana. Clim. Risk Manag., vol. 19, pp. 83-93.
- Ashraf, M., Routray, J.K. & Saeed, M., 2014. Determinants of farmers' choice of coping and adaptation measures to the drought hazard in northwest Balochistan, Pakistan. Nat. Hazard., vol. 73, no. 3, pp. 1451-73.
- Bahinipati, C.S., 2014. Determinants of farm-level adaptation diversity to cyclone and flood: insights from a farm household-level survey in Eastern India. Water Pol., vol. 17, no. 4, pp. 742-61.
- Bandiera, O. & Rasul, I., 2006. Social Networks and Technology Adoption in Northern Mozambique. The Eco. Jour., vol. 116, no. 514, pp. 869-902.
- Barros, V., Field, C., Dokken, D., Mastrandrea, M., Mach, K., Bilir, T., Chatterjee,
 M., Ebi, K., Estrada, Y. & Genova, R., 2014. IPCC, 2014: Climate Change
 2014: Impacts, adaptation, and vulnerability. Part B: Regional aspects.
 Contribution of working Group II to the fifth assessment report of the
 intergovernmental panel on climate change, Cambridge University Press,
 Cambridge.
- Bui, H.T.M. & Do, T.A., 2021. Choice of adaptation strategies to climate change among farm households in mountainous areas of Northeastern Vietnam. GeoJournal.
- Cama, M., Cristi Nicu, I., Conoscenti, C., Quénéhervé, G. & Maerker, M., 2016. The role of multicollinearity in landslide susceptibility assessment by means of Binary Logistic Regression: comparison between VIF and AIC stepwise selection. Proceedings of the pp. EPSC2016-16839.
- CARE, 2013. Climate vulnerability and capacity of ethnic minorities in the northern mountainous region of Vietnam, CARE, Canberra ACT, Australia, viewed 15 December 2017, https://careclimatechange.org/wp-content/uploads/2015/05/CARE_ClimChg_Vietnam.pdf '.

- Deressa, T.T., Hassan, R.M. & Ringler, C., 2011. Perception of and adaptation to climate change by farmers in the Nile basin of Ethiopia. The Jour. Agri. Sci., vol. 149, no. 1, pp. 23-31.
- Di Falco, S. & Veronesi, M., 2013. How Can African Agriculture Adapt to Climate Change? A Counterfactual Analysis from Ethiopia. Land Eco., vol. 89, no. 4, pp. 743-66.
- Dien, K., 2002. Population and Ethno-demography in Vietnam. Silkworm books.
- Do, T., Nguyen, C., Phung T. 2013. Assessment of Natural disasters in Vietnam's Northern Moutains. MPRA Paper No.54209. Munich Personal RePEC Archive (MPRA), p 57.
- Fahad, S. & Wang, J., 2018. Farmers' risk perception, vulnerability, and adaptation to climate change in rural Pakistan. Land Use Pol., vol. 79, pp. 301-9.
- Getahun, A.B., Ayal, D.Y., Ture, K. & Zeleke, T.T., 2021. Determinants of climate variability adaptation strategies: A case of Itang Special District, Gambella Region, Ethiopia. Clim. Serv., vol. 23, p. 100245.
- Harvey, C.A., Saborio-Rodríguez, M., Martinez-Rodríguez, M.R., Viguera, B.,
 Chain-Guadarrama, A., Vignola, R. & Alpizar, F., 2018. Climate change impacts and adaptation among smallholder farmers in Central America.
 Agri. & Food Secur., vol. 7, no. 1, p. 57.
- Huong, N.T.L., Bo, Y.S. & Fahad, S., 2017. Farmers' perception, awareness and adaptation to climate change: evidence from northwest Vietnam. Int. J. Clim. Chang. Stra. Manag. 9(4), 555-576.
- IPCC, 2014. Climate change 2014: Synthesis report, Fifth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC), Genveva, Switzerland.
- IPCC, 2018. Special report: Global warming of 1.5° C, Geneva, Switzerland: World Meteorological Organization.
- Jin, J., Wang, W. & Wang, X., 2016. Adapting agriculture to the drought hazard in rural China: household strategies and determinants. Nat. Hazd., vol. 82, no. 3, pp. 1609-19.
- Kassie, M., Jaleta, M., Shiferaw, B., Mmbando, F. & Mekuria, M., 2013. Adoption of interrelated sustainable agricultural practices in smallholder systems: Evidence from rural Tanzania. Technol. Forecast. Soci. Chang., vol. 80, no. 3, pp. 525-40.

- Khan, N.A., Gao, Q., Iqbal, M.A. & Abid, M., 2020. Modeling food growers' perceptions and behavior towards environmental changes and its induced risks: evidence from Pakistan. Environ. Sci. Pollut. Research, vol. 27, no. 16, pp. 20292-308.
- Khanal, U., Wilson, C., Hoang, V.-N. & Lee, B., 2018. Farmers' Adaptation to Climate Change, Its Determinants and Impacts on Rice Yield in Nepal. Ecol. Econ., vol. 144, pp. 139-47.
- Le Dang, H., Li, E., Bruwer, J. & Nuberg, I., 2014. Farmers' perceptions of climate variability and barriers to adaptation: lessons learned from an exploratory study in Vietnam. Mitig. Adapt. Strateg. Glob. Chang., vol. 19, no. 5, pp. 531-48.
- Li, S., Juhász-Horváth, L., Harrison, P.A., Pintér, L. & Rounsevell, M.D.A., 2017. Relating farmer's perceptions of climate change risk to adaptation behaviour in Hungary. J. Environ. Manag., vol. 185, pp. 21-30.
- Masud, M.M., Azam, M.N., Mohiuddin, M., Banna, H., Akhtar, R., Alam, A.S.A.F. & Begum, H., 2017. Adaptation barriers and strategies towards climate change: Challenges in the agricultural sector. J. Cleaner Product., vol. 156, pp. 698-706.
- Midi, H., Sarkar, S.K. & Rana, S., 2010. Collinearity diagnostics of binary logistic regression model. J. Interdisci. Math., vol. 13, no. 3, pp. 253-67.
- Mushtaq, S., Kath, J., Stone, R., Henry, R., L\u00e4derach, P., Reardon-Smith, K., Cobon, D., Marcussen, T., Cliffe, N. & Kristiansen, P., 2020. Creating positive synergies between risk management and transfer to accelerate food system climate resilience. Climat. Chang., vol. 161, no. 3, pp. 465-78.
- Ndamani, F. & Watanabe, T. 2016. Determinants of farmers adaptation to climate change: A micro level analysis in Ghana. Sci. Agric., vol. 73, pp. 201-8.
- Nelson, G.C., Rosegrant, M.W., Koo, J., Robertson, R., Sulser, T., Zhu, T., Ringler,C., Msangi, S., Palazzo, A. & Batka, M., 2009. Climate change: Impact on agriculture and costs of adaptation. Intl Food Policy Res Inst., vol. 21.
- Nelson, R., Brown, P.R., Darbas, T., Kokic, P. & Cody, K., 2007. The potential to map the adaptive capacity of Australian land managers for NRM policy using ABS data. Natural Heritage Trust.
- Ngo, Q.T., 2016. Farmers' adaptive measures to climate change induced natural shocks through past climate experiences in the Mekong River Delta, Vietnam. African J. Agric. Research, vol. 11, no. 15, pp. 1361-72.

- Nguyen Duc, K., Ancev, T. & Randall, A., 2019. Evidence of climatic change in Vietnam: Some implications for agricultural production. J. Environ. Manage., vol. 231, pp. 524-45.
- Nguyen, Y.T.B. & Leisz, S.J., 2021. Determinants of livelihood vulnerability to climate change: Two minority ethnic communities in the northwest mountainous region of Vietnam. Environ. Sci. Pol, vol. 123, pp. 11-20.
- Niles, M.T., Lubell, M. & Brown, M., 2015. How limiting factors drive agricultural adaptation to climate change. Agric. Ecos. Environ., vol. 200, pp. 178-85.
- Ojo, T.O. & Baiyegunhi, L.J.S., 2020. Determinants of climate change adaptation strategies and its impact on the net farm income of rice farmers in southwest Nigeria. Land Use Pol., vol. 95, p. 103946.
- Pachauri, R.K., Allen, M.R., Barros, V.R., Broome, J., Cramer, W., Christ, R., Church, J.A., Clarke, L., Dahe, Q. & Dasgupta, P., 2014. Climate change 2014: synthesis report. Contribution of Working Groups I, II and III to the fifth assessment report of the Intergovernmental Panel on Climate Change, Ipcc.
- PanNature, 2017. Northwest Agriculture: Identifying Challenges and Development Directions in the Context of Climate Change (in Vietnamese). Youth Publishers, Hanoi, Vietnam
- Pham, N.T.T., Nong, D. & Garschagen, M., 2019. Farmers' decisions to adapt to flash floods and landslides in the Northern Mountainous Regions of Vietnam.J. Environ. Manag., vol. 252, p. 109672.
- Pham, N.T.T., Nong, D. & Garschagen, M., 2021. Natural hazard's effect and farmers' perception: Perspectives from flash floods and landslides in remotely mountainous regions of Vietnam. Sci. Total Environ., vol. 759, p. 142656.
- Phu Yen Office of Statistics, 2018. Socio-economic report of Phu Yen district, unpublished report (in the file of Tran Thanh Van).
- Piya, L., Maharjan, K.L. & Joshi, N.P. 2013. Determinants of adaptation practices to climate change by Chepang households in the rural Mid-Hills of Nepal. Reg. Environ. Chang., vol. 13, no. 2, pp. 437-47.
- Schad, I., Schmitter, P., Saint-Macary, C., Neef, A., Lamers, M., Nguyen, L., Hilger,
 T. & Hoffmann, V., 2012. Why do people not learn from flood disasters?
 Evidence from Vietnam's northwestern mountains. Nat. Hazard., vol. 62,
 no. 2, pp. 221-41.

- Schneiderbauer, S., Fontanella Pisa, P., Delves, J.L., Pedoth, L., Rufat, S., Erschbamer, M., Thaler, T., Carnelli, F. & Granados-Chahin, S., 2021. Risk perception of climate change and natural hazards in global mountain regions: A critical review. Sci. Total Environ., vol. 784, p. 146957.
- Sertse, S.F., Khan, N.A., Shah, A.A., Liu, Y. & Naqvi, S.A.A., 2021. Farm households' perceptions and adaptation strategies to climate change risks and their determinants: Evidence from Raya Azebo district, Ethiopia. Int. J. Dis. Risk Reduct, vol. 60, p. 102255.
- Smyle, J. & Cooke, R., 2011. Vietnam: Environmental and Climate Change Assessment. Prepared for IFAD's Country Strategic Opportunities Programme 2012-2017, Retrieved from Rome, Italy.
- Soubry, B., Sherren, K. & Thornton, T.F., 2020. Are we taking farmers seriously? A review of the literature on farmer perceptions and climate change, 2007– 2018. J. Rural Stud., vol. 74, pp. 210-22.
- Thanh Thi Pham, N., Nong, D., Raghavan Sathyan, A. & Garschagen, M. 2020. Vulnerability assessment of households to flash floods and landslides in the poor upland regions of Vietnam. Climat. Risk Manag., vol. 28, p. 100215.
- Thornton, T.F. & Comberti, C., 2017. Synergies and trade-offs between adaptation, mitigation and development. Climat. Chang., vol. 140, no. 1, pp. 5-18.
- Tran, V.T., An-Vo, D.-A., Cockfield, G. & Mushtaq, S., 2021. Assessing Livelihood Vulnerability of Minority Ethnic Groups to Climate Change: A Case Study from the Northwest Mountainous Regions of Vietnam. Sustainability, vol. 13, no. 13, p. 7106.
- Tran, V.T., An-Vo, D.-A., Cockfield, G. & Mushtaq, S., 2022. Nuanced assessment of livelihood resilience through the intersectional lens of gender and ethnicity: Evidence from small-scale farming communities in the upland regions of Vietnam. J. Rural Stud., vol. 92, pp. 68-78.
- Trinh, T.Q., Rañola, R.F., Camacho, L.D. & Simelton, E., 2018. Determinants of farmers' adaptation to climate change in agricultural production in the central region of Vietnam. Land Use Pol., vol. 70, no. Supplement C, pp. 224-31.
- UNDP, 2015. Viet Nam special report on managing the risks of extreme events and disasters to advance climate change adaptation. UNDP, Vietnam. viewed at 25 June 2018, http://www.vn.undp.org/content/vietnam/en/home/library/environment

92

_climate/viet_nam_special_report_on_managing_the_risks_of_extreme_e vents_and_disasters.html>.

- Van, S.T., Boyd, W., Slavich, P. & Van, T.M., 2015. Perception of climate change and farmers' adaptation: A case study of poor and non-poor farmers in northern central coast of Vietnam. J. Basic and Applied Sci., vol. 11, pp. 323-42.
- Vien, T.D., 2003. Culture, Envionment, and Farming Systems in Vietnam's Northern Mountain Region. Japanese J. Southeast Asian Stud., vol. 41, no. 2, pp. 180-205.
- Wheeler, S., Zuo, A. & Bjornlund, H., 2013. Farmers' climate change beliefs and adaptation strategies for a water scarce future in Australia. Glo. Environ. Chang., vol. 23, no. 2, pp. 537-47.
- Wheeler, T. & Braun, J.V., 2013. Climate Change Impacts on Global Food Security. Science, vol. 341, no. 6145, pp. 508-13.

This chapter provided an overview of farmers' perceptions of changes in climate variability and related extreme events, their responses to these changes and main factors affecting farmers' adaptation choices. Local farmers have observed a recent increase in temperature and extreme weather events, but a decrease in rainfall. To adapt to climate change and variability, farmers adopted several adaptation practices such as changing crop varieties, crop diversification, crop management and protection, and finding off-farm jobs. By employing the binary logistic model, we discovered that social-economic factors (especially ethnic location), farm size and institutional factors plays an important role in adopting adaptive practices to cope with climate change and variability.

CHAPTER 6: CONCLUSIONS

This study focused on the analysis of the relationship between vulnerability, resilience, and adaptation of rural farmers to climate change in Phu Yen district, Son La province, based on the data from a survey of 240 households from four ethnic communities. The picture of vulnerability at community level under the impacts of climate change was derived by applying the composite LVI and a substitute approach (LVI- IPCC), in combination with comprehensive qualitative data. In addition, an intersectional approach was deployed for a nuanced view of subjective resilience at household level based on individual's self-assessments. Finally, the study identified some of the drivers of farmers' adaptation choices. The findings of the research provide information and insights to help policy makers alleviate poverty and increase livelihood resilience, especially in disadvantaged regions, such as mountainous regions, and disadvantaged groups such as ethnic minorities. This chapter will briefly summarise and review the important results and identify the significant contributions and implications of this study. Finally, the chapter concludes by providing suggestions for policy makers and future researchers.

6.1 Summary of important findings

This study is the first assessment of the vulnerability at the community level comparing different ethnic groups in the Northwest Mountainous Regions. The groups were Thai, Muong, Dao, and Hmong ethnic groups. By applying the livelihood vulnerability index framework (composite LVI and LVI-IPCC), we identified the differences in vulnerability levels. The overall indices revealed differing vulnerabilities by ethnicity and location, with Hmong and Dao being the most vulnerable groups. These groups are more vulnerable in terms of low education levels, diversity of income sources, agricultural training, organisational membership, access to health services and water resources, and housing location. The study then deployed a nuanced assessment of intersectionality of gender and ethnicity to explore how perception of household livelihood resilience differ among selected ethnic communities. Our analysis suggests that ethnicity was relatively more important than gender in explaining the variations, but within ethnic and locational differences there are nested gender differences. Men generally reported higher resilience scores than women did but there is significant variation in average scores across selected ethnic communities. The results also suggest a number of factors likely to contribute to better livelihood resilience self-assessment, including education, wage-paying jobs, agricultural trainings, crop diversification, social membership, roads, and irrigation systems.

In relation to climate change awareness and perceptions, the overwhelming majority of farmers had noticed changes in key variables and weather extremes events, though these differed somewhat across the groups. In response, local farmers have undertaken some adaptation measures, especially amongst those with more resources. The most recorded adaption practices are crop management and protection (soil and plant), crop diversification, finding off-farm jobs and changing crop varieties. Limitations to adaptation include financial restrictions, farming features, lack of family labour and weather information are significant impediments hindering the effectiveness of adaptation strategies. Binary logistic models were used to further examine the factors affecting farmers' choices to adapt to climate change and climate-induced hazards are analysed. The resulting adaption models suggest that age, ethnicity, farming size, access to irrigation, extension services are most important factors deciding specific adaptation choices. Other factors are levels of education, gender, credit availability, access to climate information and group membership also influence farmers' decisions to adapt to climate change and climate hazards.

6.2 Significance and scientific contribution of the study

This research aimed to be a comprehensive assessment of vulnerability at community level that could inform policies to reduce vulnerability to a changing climate. The results reveal ethnicity is the more important factor than gender in determining household livelihood capitals supporting livelihood resilience in the study regions. The research results indicate an agreement between farmers' perception on temperature with the actual trend; however, farmers' perception on rainfall are somewhat at variance with the climate data. This implies there is a gap in perceptions of climate variability among ethnic communities with the trend of the climate variables. Additionally, this study found that ethnic location, farmers' socioeconomic, farming features, and institutional conditions are the main contributing factors strongly influencing their adaptation choices to climate change. These findings will support decision-making and planning by local government, crop industries, and ethnic united organisation. For example, feasible direction and policies that enhance better farmers' knowledge through investment in education systems and the quality of extension services focusing on cultivation and breeding techniques, and better communication with farmers.

This study contributed to a regional illustration of assessing vulnerability at community and perceived household livelihood resilience assessment to inform future studies across other regions and countries that can support adaptive strategies in similar communities elsewhere.

Finally, this work aligns well with other work in the climate change impacts research and will support policy makers to reduce vulnerability, enhance farmers' perceptions and strengthen households' adaptation to climate change and climate hazards.

6.3 Recommendations

6.3.1 Policy recommendations

Based on the empirical findings, this study suggests several policy interventions that could assist to reduce the household vulnerability, to enhance local farmers resilience and for better adaptation process towards climate change and climate hazards, such as:

- Enhancing farmers' literacy and organizing vocational training courses.

- Strengthening the quality of agricultural extension services (both extension staff and training documents).

- Facilitating income diversification strategies by trainings and concessional loans.

- Policy makers should consider gender and minority ethnic group as important factors in future projects regarding resilience enhancement, especially for women and amongst disadvantaged ethnic communities.

Other recommendations include improving the quality of drinking water sources, upgrading irrigation systems, providing agricultural inputs with reasonable costs, and providing timely local weather forecast, should also be emphasized, and given more attention from local governments.

6.3.2 Recommendations for future research

This current work investigates vulnerability at community level and factors affecting households' vulnerability to climate change. Future studies should consider analysing which livelihoods are suitable and able to support local people to reduce their vulnerability to climate change.

This study measures household livelihood resilience based on the quantitative survey with assuming all indicators have equal weighting to livelihood perceived resilience. The indicators may not capture essential variables, such as indigenous knowledge. Future research should involve research participants weighting the vulnerability and resilience indicators, which helps researchers better understand the importance of various indicators. Future evaluations could investigate which adaptations result in greater economic efficiency, as this will help farmers become more resilient to future climate change. Future studies may also consider analysing farmer adaptation in a more comprehensive way, considering changes in the institutional, environmental, and economic conditions. Furthermore, the data should be collected at different times (panel data), allowing researchers to compare and examine the study areas in depth.

REFERENCES

- Adger, WN 2000. Social and ecological resilience: are they related?. Progress in human geography, vol. 24, no. 3, pp. 347-64.
- Adger, WN 2006. Vulnerability. Global Environmental Change vol. 16, no. 3, pp. 268-81.
- Adger, WN & Kelly, PM 1999. Social vulnerability to climate change and the architecture of entitlements. Mitigation and adaptation strategies for global change, vol. 4, no. 3, pp. 253-66.
- Adger, WN, Agrawala, S, Mirza, MMQ, Conde, C, o'Brien, K, Pulhin, J, Pulwarty, R, Smit, B & Takahashi, K 2007. Assessment of adaptation practices, options, constraints and capacity. Climate change, pp. 717-43.
- Adger, WN & Barnett, J 2000. Four reasons for concern about adaptation to climate change. Environment and Planning A, vol. 41, no. 11, pp. 2800.5.
- Alam, GMM, Alam, K, Mushtaq, S & Filho, WL 2018. How do climate change and associated hazards impact on the resilience of riparian rural communities in Bangladesh? Policy implications for livelihood development', Environmental Science & Policy, vol. 84, pp. 7-18.
- Anh, NT 2016. Effects of Natural Disasters on Agricultural Production Activities in the Cambodia-Laos- Vietnam Development Triangle Area: Case Studies of Ratanakiri (Cambodia), Attapeu (Laos) and Kon Tum (Vietnam) Provinces.
 S. Rajaratnam School of International Studies.
- Asafu-Adjaye, J 2014. The Economic Impacts of Climate Change on Agriculture in Africa. Journal of African Economies, vol. 23, no. suppl_2, pp. ii17-ii49.
- Babel, MS, Agarwal, A, Swain, DK & Herath, S 2011. Evaluation of climate change impacts and adaptation measures for rice cultivation in Northeast Thailand. Climate Research, vol. 46, no. 2, pp. 137-46.
- Bank, W 2017. Country Partnership Framework for the Socialist Republic of Vietnam, for the Period FY18-FY22.
- Bardsley, DK & Sweeney, SM 2010. Guiding climate change adaptation within vulnerable natural resource management systems. Environmental Management, vol. 45, no. 5, pp. 1127-41.

Beaumont, LJ, Pitman, A, Perkins, S, Zimmermann, NE, Yoccoz, NG & Thuiller, W 2011. Impacts of climate change on the world's most exceptional ecoregions. Proceedings of the national academy of sciences, p. 201007217.

Benson, C 2017. The economic Impact of Natural Disasters in Viet Nam.

- Berkes, F & Folke, C 1998. Linking social and ecological systems for resilience and sustainability. Linking social and ecological systems: management practices and social mechanisms for building resilience, vol. 1, no. 4, pp. pp.1-14.
- Berkes, F & Jolly, D 2001. Adapting to Climate Change: Social-Ecological Resilience in a Canadian Western Arctic Community. Conservation Ecology, vol. 5(2):-18.
- Berkes, F, J. Colding & Folke, C 2003. Navigating social-ecological systems: Building resilience for complexity and change. Cambridge University Press, Cambridge, UK.
- Birkmann, J 2006. Measuring vulnerability to promote disaster-resilient societies: Conceptual frameworks and definitions. Measuring vulnerability to natural hazards: Towards disaster resilient societies, vol. 1, pp. 9-54.
- Blanc, É 2012. The impact of climate change on crop yields in Sub-Saharan Africa. American Journal of Climate Change, vol. 1, no. 01, p. 1.
- Brooks, N 2003. Vulnerability, Risk and Adaptation: A Conceptual Framework. Tyndall Centre for Climate Change Working Paper 38, University of East Anglia, Norwich, UK
- Brown, K 2013. Global environmental change I: A social turn for resilience?. Progress in human geography, vol. 38, no. 1, pp. 107-17.
- Bruneau, M, Chang, SE, Eguchi, RT, Lee, GC, O'Rourke, TD, Reinhorn, AM, Shinozuka, M, Tierney, K, Wallace, WA & Von Winterfeldt, D 2003. A framework to quantitatively assess and enhance the seismic resilience of communities. Earthquake spectra, vol. 19, no. 4, pp. 733-52.
- Burke, M & Emerick, K 2016. Adaptation to Climate Change: Evidence from US Agriculture. American Economic Journal: Economic Policy, vol. 8, no. 3, pp. 106-40.

- Burney, J & Ramanathan, V 2014. Recent climate and air pollution impacts on Indian agriculture. Proceedings of the national academy of sciences, vol. 111, no. 46, pp. 16319-24.
- Burton, I, Huq, S, Lim, B, Pilifosova, O & Schipper, EL 2002. From impacts assessment to adaptation priorities: the shaping of adaptation policy. Climate policy, vol. 2, no. 2-3, pp. 145-59.
- Cardona, OD, M.K. van Aalst, J. Birkmann, M. Fordham, G. McGregor, R. Perez, R.S. Pulwarty, E.L.F. Schipper & Sinh, BT 2012. Determinants of risk: exposure and vulnerability. In: Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptationn . [Field, C.B., V. Barros, T.F. Stocker, D. Qin, D.J. Dokken, K.L. Ebi, M.D. Mastrandrea, K.J. Mach, G.-K. Plattner, S.K. Allen, M. Tignor, and P.M. Midgley (eds.)]. A Special Report of Working Groups I and II of the Intergovernmental Panel on Climate Change (IPCC). no. Cambridge University Press, Cambridge, UK, and New York, NY, USA, pp. pp. 65-108.
- CARE 2013. Climate vulnerability and capacity of ethnic minorities in the northern mountainous region of Vietnam. CARE, Canberra ACT, Australia, viewed 15 December 2017, https://careclimatechange.org/wpcontent/uploads/2015/05/CARE_ClimChg_Vietnam.pdf '.
- Carpenter, S, Walker, B, Anderies, JM & Abel, N 2001. From metaphor to measurement: resilience of what to what?. Ecosystems, vol. 4, no. 8, pp. 765-81.
- Chambers, R & Conway, G 1992. Sustainable rural livelihoods: practical concepts for the 21st century. Institute of Development Studies (UK).
- Chen, S, Chen, X & Xu, J 2016. Impacts of climate change on agriculture: Evidence from China. Journal of Environmental Economics and Management, vol. 76, pp. 105-24.
- Constas, Mea 2014. A common analytical model for resilience management causal framework and methodological options. Resilience Measurement Technical Working Group, FSiN Technical Series Paper No. 2, World Food Program and Food and Agriculture Organization, p. 52pg.

- Cutter, SL, Barnes, L, Berry, M, Burton, C, Evans, E, Tate, E & Webb, J 2008. A place-based model for understanding community resilience to natural disasters. Global Environmental Change, vol. 18, no. 4, pp. 598-606.
- Dasgupta, S, Laplante, B, Meisner, C, Wheeler, D & Yan, J 2009. The impact of sea level rise on developing countries: a comparative analysis. Climatic Change, vol. 93, no. 3, pp. 379-88.
- Davidson, DJ 2010. The applicability of the concept of resilience to social systems: some sources of optimism and nagging doubts. Society and Natural Resources, vol. 23, no. 12, pp. 1135-49.
- Deressa, T, Hassan, RM & Ringler, C 2008. Measuring Ethiopian farmers' vulnerability to climate change across regional states. Environmental and Production Technology Division, International Food Policy Research Institute.
- Desch, xea, nes, O & Greenstone, M 2012. The Economic Impacts of Climate Change: Evidence from Agricultural Output and Random Fluctuations in Weather: Reply. The American Economic Review, vol. 102, no. 7, pp. 3761-73.
- Do T, NC, Phung T 2013. Assessment of Natural disasters in Vietnam's Northern Moutains. MPRA Paper No.54209. Munich Personal RePEC Archive (MPRA), p 57.
- Duit, A, Galaz, V, Eckerberg, K & Ebbesson, J 2010. Governance, complexity, and resilience. Global Environmental Change, vol. 20, no. 3, pp. 363-8.
- Eakin, H & Luers, AL 2006. Assessing the vulnerability of social-environmental systems. Annual review of environment and resources, vol. 31,18 June, pp. 365-94.
- Eckstein, D, Künzel, V & Schäfer, L 2017. Global Climate Risk Index 2018. Germanwatch, Bonn, Germany, view 20 December 2010, <https://germanwatch.org/en/download/20432.pdf>.
- Engle, NL 2011. Adaptive capacity and its assessment. Global Environmental Change, vol. 21, no. 2, pp. 647-56.

- Eriksen, SH, Brown, K & Kelly, PM 2005. The dynamics of vulnerability: locating coping strategies in Kenya and Tanzania. The geographical journal, vol. 171, no. 4, pp. 287-305.
- Few, R & Tran, PG 2010. Climatic hazards, health risk and response in Vietnam: Case studies on social dimensions of vulnerability. Global Environmental Change, vol. 20, no. 3, pp. 529-38.
- Fisher, AC, Hanemann, WM, Roberts, MJ & Schlenker, W 2012. The Economic Impacts of Climate Change: Evidence from Agricultural Output and Random Fluctuations in Weather: Comment. The American Economic Review, vol. 102, no. 7, pp. 3749-60.
- Folke, C 2006. Resilience: The emergence of a perspective for social–ecological systems analyses. Global Environmental Change, vol. 16, no. 3, pp. 253-67.
- Folke, C, Carpenter, S, Elmqvist, T, Gunderson, L, Holling, CS & Walker, B 2002.
 Resilience and sustainable development: building adaptive capacity in a world of transformations. AMBIO: A journal of the human environment, vol. 31, no. 5, pp. 437-40.
- Ford, JD & Smit, B 2004. A framework for assessing the vulnerability of communities in the Canadian Arctic to risks associated with climate change. Arctic, pp. 389-400.
- Ford, JD, Smit, B, Wandel, J, Allurut, M, Shappa, K, Ittusarjuat, H & Qrunnut, K 2008. Climate change in the Arctic: current and future vulnerability in two Inuit communities in Canada. The geographical journal, vol. 174, no. 1, pp. 45-62.
- Füssel, H-M 2007. Vulnerability: A generally applicable conceptual framework for climate change research. Global Environmental Change, vol. 17, no. 2, pp. 155-67.
- Füssel, H-M & Klein, RJ 2006. Climate change vulnerability assessments: an evolution of conceptual thinking. Climatic Change, vol. 75, no. 3, pp. 301-29.
- Gallopín, GC 2006. Linkages between vulnerability, resilience, and adaptive capacity. Global Environmental Change, vol. 16, no. 3, pp. 293-303.

GoV 2008. National Target Programme on responding to climate change.

- GSO 2015. Results of Vietnam household living standard survey 2014. Hanoi: General Statistical Office.
- Ha, TT 2007. The impacts of the Doi Moi process on communities and forest land management in three Northeastern provinces in the Northern Mountainous Region of Vietnam. The Australian National University, Canberra.
- Holling, CS 1973. Resilience and stability of ecological systems. Annual review of ecology and systematics, vol. 4, no. 1, pp. 1-23.
- Houser, T, Hsiang, S, Kopp, R & Larsen, K 2015. Economic risks of climate change: an American prospectus. Columbia University Press, New York, USA.
- Huynh, LTM & Stringer, LC 2018. Multi-scale assessment of social vulnerability to climate change: An empirical study in coastal Vietnam. Climate Risk Management, vol. 20, pp. 165-80.
- Imai, KS, Gaiha, R & Kang, W 2011. Poverty, inequality and ethnic minorities in Vietnam. International Review of Applied Economics, vol. 25, no. 3, pp. 249-82.
- IMHEN & UNDP 2015. Vietnam Special Report on Managing Risks of Extreme Events and Disasters to Advance Climate Change Adaptation Hanoi, Vietnam.
- IPCC 2001. Climate change 2001: Impacts, adaptation and vulnerability. Contribution of Working Group II to the Third Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge.
- IPCC 2007. Summary for policy-makers, in Climate change 2007: The physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK.
- IPCC 2013. Climate Change 2013: The Physical Science Basis, Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.

- IPCC 2014. Climate change 2014: Synthesis report, Fifth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC). Genveva, Switzerland.
- Jones, L & Tanner, T 2015. Measuring'subjective resilience': using peoples' perceptions to quantify household resilience (Working Paper No.423), vol.23. Overseas Development Institute Working.
- Kelly, PM & Adger, WN 2000. Theory and practice in assessing vulnerability to climate change andFacilitating adaptation. Climatic Change, vol. 47, no. 4, pp. 325-52.
- Kohler, T & Maselli, D 2012. Mountains and Climate Change: From Understanding to Action, third ed., Geographica Bernensia with support of the Swiss Agency for Development and Cooperation (SDC) and an International Team of Contributors.Bern, Switzerland.
- Krishnan, P, Swain, D, Bhaskar, BC, Nayak, S & Dash, R 2007. Impact of elevated
 CO 2 and temperature on rice yield and methods of adaptation as evaluated
 by crop simulation studies. Agriculture, ecosystems & environment, vol.
 122, no. 2, pp. 233-42.
- Kumar, KK & Tholkappian, S 2006. Relative vulnerability of Indian coastal districts to sea-level rise and climate extremes. International review for environmental strategies, vol. 6, no. 1, pp. 3-22.
- Kurukulasuriya, P & Rosenthal, S 2013. Climate change and agriculture: A review of impacts and adaptations (English). In: Paper No.91 in Climate Change Series. Washington DC, World Bank, view at 20 January 2018, http://documents.worldbank.org/curated/en/757601468332407727/Climate-change-and-agriculture-a-review-of-impacts-and-adaptations>.
- Le Dang, H, Li, E, Bruwer, J & Nuberg, I 2014. Farmers' perceptions of climate variability and barriers to adaptation: lessons learned from an exploratory study in Vietnam. Mitigation and adaptation strategies for global change, vol. 19, no. 5, pp. 531-48.
- Lebel, L, Anderies, J, Campbell, B, Folke, C, Hatfield-Dodds, S, Hughes, T & Wilson, J 2006. Governance and the capacity to manage resilience in regional socialecological systems. Ecology and Society, vol. 11, no. 1.

- Leichenko, RM & O'brien, KL 2002. The dynamics of rural vulnerability to global change: the case of southern Africa. Mitigation and adaptation strategies for global change, vol. 7, no. 1, pp. 1-18.
- Lemos, MC, Boyd, E, Tompkins, E, Osbahr, H & Liverman, D 2007. Developing adaptation and adapting development. Ecology and Society, vol. 12, no. 2, pp. 26-30.
- Lobell, DB, Hammer, GL, McLean, G, Messina, C, Roberts, MJ & Schlenker, W 2013. The critical role of extreme heat for maize production in the United States. Nature Climate Change, vol. 3, p. 497.
- Lynn, K, MacKendrick, K & Donoghue, EM 2011. Social vulnerability and climate change: synthesis of literature. Gen. Tech. Rep. PNW-GTR-838. Portland, OR: US Department of Agriculture, Forest Service, Pacific Northwest Research Station. 70 p., vol. 838.
- Maharjan, K.L., Joshi, N.P., 2013. Climate change, Agriculture and Rural Livelihoods in Developing Countries. Springer, Japan.
- Maiti, S, Jha, SK, Garai, S, Nag, A, Chakravarty, R, Kadian, K, Chandel, B, Datta,
 K & Upadhyay, R 2015. Assessment of social vulnerability to climate change in the eastern coast of India. Climatic Change, vol. 131, no. 2, pp. 287-306.
- Manyena, SB 2006. The concept of resilience revisited. Disasters, vol. 30, no. 4, pp. 434-50.
- Masud, MM, Rahman, MS, Al-Amin, AQ, Kari, F & Leal Filho, W 2014. Impact of climate change: an empirical investigation of Malaysian rice production.
 Mitigation and adaptation strategies for global change, vol. 19, no. 4, pp. 431-44.
- McLaughlin, P & Dietz, T 2008. Structure, agency and environment: Toward an integrated perspective on vulnerability. Global Environmental Change, vol. 18, no. 1, pp. 99-111.
- Mestre-Sanchís, F & Feijóo-Bello, ML 2009. Climate change and its marginalizing effect on agriculture. Ecological economics, vol. 68, no. 3, pp. 896-904.
- MONRE 2016. Climate change and sea level rise scenarios for Viet Nam. Minisry of Natural Resources and Environment, Hanoi.
- MONRE 2017. National Disaster Risk in Vietnam in the Period 2006-2016 and forecasting and warning warning system.
- Moreno, A & Becken, S 2009. A climate change vulnerability assessment methodology for coastal tourism. Journal of Sustainable Tourism, vol. 17, no. 4, pp. 473-88.
- Moser, CO 1998. The asset vulnerability framework: reassessing urban poverty reduction strategies. World Development, vol. 26, no. 1, pp. 1-19.
- Nam, PK 2011. Social capital and private adaptation to climate change: Evidence from the Mekong River Delta in Vietnam. Institute for International Economic Policy (IIEP) conference on the "Economics of Adaptation to Climate Change," at George Washington University, Washington, DC.
- Nelson, R, Kokic, P, Crimp, S, Martin, P, Meinke, H, Howden, S, de Voil, P & Nidumolu, U 2010. The vulnerability of Australian rural communities to climate variability and change: Part II—Integrating impacts with adaptive capacity. Environmental Science & Policy, vol. 13, no. 1, pp. 18-27.
- Nguyen, DC, Vo, HT & Chu, TH 2013. Application of Livelihood Vulnerability Index to Assess Risks from Flood Vulnerability and Climate Variability—A Case Study in the Mekong Delta of Vietnam. Journal of Environmental Science and Engineering. Formerly part of Journal of Environmental Science and Engineering.
- Nguyen, YTB & Leisz, SJ 2021. Determinants of livelihood vulnerability to climate change: Two minority ethnic communities in the northwest mountainous region of Vietnam. Environmental Science & Policy, vol. 123, pp. 11-20.
- Nyong, A, Dabi, D, Adepetu, A, Berthe, A & Ihemegbulem, V 2009. Vulnerability in the sahelian zone of northern Nigeria: a household-level assessment, in CC Neil Leary, Jyoti Kulkarni, Juan Pulhin, Anthony Nyong (ed.), Climate Change and Vulnerability. Routledge, Earthscan in UK and USA, pp. 218-38.
- O'Brien, K 2004a. What's in a Word?: Conflicting Interpretations of Vulnerability in Climate Change Research. Center for International Climate and Environmental Research, Blinder, Oslo, Norway.
- O'Brien, K, Leichenko, R, Kelkar, U, Venema, H, Aandahl, G, Tompkins, H, Javed, A, Bhadwal, S, Barg, S & Nygaard, L 2004b. Mapping vulnerability to

multiple stressors: climate change and globalization in India. Global Environmental Change, vol. 14, no. 4, pp. 303-13.

- O'Brien, K, Sygna, L, Leichenko, R, Adger, WN, Barnett, J, Mitchell, T, Schipper, L, Tanner, T, Vogel, C & Mortreux, C 2008. Disaster risk reduction, climate change adaptation and human security. Report prepared for the Royal Norwegian Ministry of Foreign Affairs by the Global Environmental Change and Human Security Project. GECHS IPO, Oslo, Norway.
- Paton, D 2006. Disaster resilience: building capacity to co-exist with natural hazards and their consequences, in D D. Paton, Johnston (ed.). Disaster resilience: An integrated approach, Charles C Thomas Publisher, Illinois, pp. 3-10.
- Paton, D & Johnston, D 2017. Disaster resilience: an integrated approach. Charles C Thomas Publisher.
- Quandt, A 2018. Measuring livelihood resilience: The Household Livelihood Resilience Approach (HLRA). World Development, vol. 107, pp. 253-63.
- Renaud, FG, Birkmann, J, Damm, M & Gallopín, GC 2010. Understanding multiple thresholds of coupled social–ecological systems exposed to natural hazards as external shocks. Natural hazards, vol. 55, no. 3, pp. 749-63.
- Robards, MD, Schoon, ML, Meek, CL & Engle, NL 2011. The importance of social drivers in the resilient provision of ecosystem services. Global Environmental Change, vol. 21, no. 2, pp. 522-9.
- Roberts, MJ, Schlenker, W & Eyer, J 2013. Agronomic Weather Measures in Econometric Models of Crop Yield with Implications for Climate Change. American Journal of Agricultural Economics, vol. 95, no. 2, pp. 236-43.
- Roudier, P, Sultan, B, Quirion, P & Berg, A 2011. The impact of future climate change on West African crop yields: What does the recent literature say?. Global Environmental Change, vol. 21, no. 3, pp. 1073-83.
- Salack, S, Sarr, B, Sangare, S, Ly, M, Seidou Sanda, I & Kunstmann, H 2015. Crop-climate ensemble scenarios to improve risk assessment and resilience in the semi-arid regions of West Africa. Climate research, vol. 65, pp. 107-121.

- Seidl, R, Rammer, W & Lexer, MJ 2011. Adaptation options to reduce climate change vulnerability of sustainable forest management in the Austrian Alps. Canadian Journal of Forest Research, vol. 41, no. 4, pp. 694-706.
- Shah, KU, Dulal, HB, Johnson, C & Baptiste, A 2013. Understanding livelihood vulnerability to climate change: Applying the livelihood vulnerability index in Trinidad and Tobago. Geoforum, vol. 47, pp. 125-37.
- Shrestha, S, Gyawali, B & Bhattarai, U 2013. Impacts of climate change on irrigation water requirements for rice–wheat cultivation in Bagmati River Basin, Nepal. Journal of Water and Climate Change, vol. 4, no. 4, pp. 422-39.
- Skoufias, E, Rabassa, M, Olivieri, S & Brahmbhatt, M 2011. The Poverty Impacts of Climate Change. Economic Premise, No. 51. World Bank, Washington, DC, viewed at 28 June 2018, < https://openknowledge.worldbank.org/handle/10986/10102>.
- Smit, B & Pilifosova, O 2003. Adaptation to climate change in the context of sustainable development and equity: Climate change 2001: Impacts, Adaptation, and Vulnerability. Cambrigde University Press, New York, USA.
- Smit, B & Wandel, J 2006. Adaptation, adaptive capacity and vulnerability. Global Environmental Change, vol. 16, no. 3, pp. 282-92.
- Tanner, T, Lewis, D, Wrathall, D, Bronen, R, Cradock-Henry, N, Huq, S, Lawless,
 C, Nawrotzki, R, Prasad, V, Rahman, MA, Alaniz, R, King, K, McNamara, K,
 Nadiruzzaman, M, Henly-Shepard, S & Thomalla, F 2014. Livelihood resilience in the face of climate change. Nature Climate Change, vol. 5, p. 23.
- Tierney, K & Bruneau, M 2007. Conceptualizing and measuring resilience: A key to disaster loss reduction. TR news, no. 250 (May-June), pp. 14-7.
- Tran, TQ, Nguyen, SH, Vu, HV & Nguyen, VQ 2015. A note on poverty among ethnic minorities in the Northwest region of Vietnam. Post-Communist Economies, vol. 27, no. 2, pp. 268-81.
- Trinh, TQ, Rañola, RF, Camacho, LD & Simelton, E 2018. Determinants of farmers' adaptation to climate change in agricultural production in the central region of Vietnam. Land Use Policy, vol. 70, no. Supplement C, pp. 224-31.

- Tschakert, P 2007. Views from the vulnerable: understanding climatic and other stressors in the Sahel. Global Environmental Change, vol. 17, no. 3, pp. 381-96.
- Turner, BL, Kasperson, RE, Matson, PA, McCarthy, JJ, Corell, RW, Christensen, L, Eckley, N, Kasperson, JX, Luers, A & Martello, ML 2003a. A framework for vulnerability analysis in sustainability science. Proceedings of the national academy of sciences, vol. 100, no. 14, pp. 8074-9.
- Turner, BL, Matson, PA, McCarthy, JJ, Corell, RW, Christensen, L, Eckley, N, Hovelsrud-Broda, GK, Kasperson, JX, Kasperson, RE & Luers, A 2003b. Illustrating the coupled human–environment system for vulnerability analysis: three case studies. Proceedings of the national academy of sciences, vol. 100, no. 14, pp. 8080-5.
- Tuyen, TQ 2016. Income sources and inequality among ethnic minorities in the Northwest region, Vietnam. Environment, Development and Sustainability, vol. 18, no. 4, pp. 1239-54.
- UNDP 2015. Viet Nam special report on managing the risks of extreme events and disasters to advance climate change adaptation. UNDP, Vietnam, viewed at 25 June 2018, .
- Van, ST, Boyd, W, Slavich, P & Van, TM 2015. Perception of climate change and farmers' adaptation: A case study of poor and non-poor farmers in northern central coast of Vietnam. Journal of Basic and Applied Sciences, vol. 11, pp. 323-42.
- Walker, B, Holling, CS, Carpenter, S & Kinzig, A 2004. Resilience, adaptability and transformability in social–ecological systems. Ecology and Society, vol. 9, no. 2, pp. 158-67.
- Walker, B, Gunderson, L, Kinzig, A, Folke, C, Carpenter, S & Schultz, L 2006. A handful of heuristics and some propositions for understanding resilience in social-ecological systems. Ecology and Society, vol. 11, no. 1, pp. 256-71.

- Watson, RT, Zinyowera, MC & Moss, RH 1996. Climate change 1995. Impacts, adaptations and mitigation of climate change: scientific-technical analyses.
- Weiss, J 2009. The Economics of Climate Change in Southeast Asia: A Regional Review. Asian Development Bank, Manila, Phiphipins, viewed 15 December 2017, <http://hdl.handle.net/11540/179>.
- Welch, JR, Vincent, JR, Auffhammer, M, Moya, PF, Dobermann, A & Dawe, D 2010. Rice yields in tropical/subtropical Asia exhibit large but opposing sensitivities to minimum and maximum temperatures. Proceedings of the national academy of sciences, vol. 107, no. 33, p. 14562.
- World Bank 2010. The social dimensions of adaptation to climate change in Vietnam. Discussion Paper number 12 December 2010. Economics of Adaptation to Climate Change, Washington, D.C.
- Zhai, F & Zhuang, J 2012. Agricultural impact of climate change: A general equilibrium analysis with special reference to Southeast Asia, in MB Venkatachalam Anbumozhi, Selvarajah Pathmarajah, Vangimalla R Reddy (ed.). Climate Change in Asia and the Pacific: How Can Countries Adapt, SAGE Publications India, pp. 17-35.